Contents lists available at ScienceDirect



journal homepage: www.elsevier.com/locate/eor

# Invited Review in Celebration of the 50th Anniversary of EURO

# Fifty years of portfolio optimization

# Ahti Salo<sup>a,\*</sup>, Michalis Doumpos<sup>c</sup>, Juuso Liesiö<sup>b</sup>, Constantin Zopounidis<sup>c,d</sup>

<sup>a</sup> Department of Mathematics and Systems Analysis, Aalto University School of Science, Finland

<sup>b</sup> Department of Information and Service Management, Aalto University School of Business, Finland

<sup>c</sup> Financial Engineering Laboratory, School of Production Engineering and Management, Technical University of Crete, Greece

<sup>d</sup> Audencia Business School, France

SEVIER

# ARTICLE INFO

Keywords: Portfolio optimization Multiattribute value/utility theory Multicriteria decision aiding Portfolio decision analysis Resource allocation

# ABSTRACT

The allocation of resources to alternative investment opportunities is one of the most important decisions organizations and individuals face. These decisions can be guided by building and solving portfolio optimization models that capture the salient aspects of the investment problem, including decision-makers' preferences, multiple objectives, and decision opportunities over the planning horizon. In this paper, we give a historically grounded overview of portfolio optimization which, as a field within operational research with roots in finance, is vast thanks to many decades of research and the huge diversity of problems that have been tackled. In particular, we provide a unified and therefore unique treatment that covers the full breadth of portfolio optimization problems, including, for instance, the allocation of resources to financial assets and the selection of indivisible assets such as R&D projects. We also identify opportunities for future methodological and applied research, hoping to inspire researchers to contribute to the growing field of portfolio optimization.

# 1. Introduction

In a broad sense, portfolio management refers to all the processes of identifying, selecting, and managing a collection of investment assets by an organization or an individual. These processes involve the screening and selection of these assets; the allocation of resources to those selected; the monitoring of portfolio performance over time; and the updating of the portfolio by introducing new assets or discontinuing earlier allocations, for instance. In this context, determining the 'best' composition of the portfolio is a crucial decision problem that can be addressed through portfolio optimization.

The origins of portfolio optimization can be traced to the pivotal contributions by Markowitz (1952). Since the establishment of this modern portfolio theory, the field of portfolio optimization has become immense. Presently, it spans a wealth of analytical and often sophisticated models that build on many quantitative disciplines. Accordingly, portfolio optimization can be approached from complementary perspectives. To begin with, pivotal contributions in decision theory, such as the representation of preferences with the Neumann–Morgenstern utility function, allow portfolio optimization problems to be addressed within the framework of expected utility maximization and its variants. Second, probabilistic and statistical studies focus on deriving probability distributions over future outcomes so that the consequences of the selected portfolio for the decision-maker can be assessed. Third, advances in computational capabilities help in formulating tractable

optimization models and designing, implementing, and evaluating efficient algorithms for solving these. Fourth, concerns in leveraging effective decision support include, among others, the need to ensure that the portfolio model constitutes a valid problem representation and that interactions between the model and its users are aligned with the model's intended purpose.

As portfolio optimization originated some 70 years ago, we have not limited ourselves to the past half a century mentioned in the title. Instead, we highlight selected milestones in the evolution of portfolio optimization since its birth. This notwithstanding, most emphasis is given to developments from the past twenty years or so, as this allows us to give an account of the current state-of-the-art in portfolio optimization. We hope this account and our reflections on future opportunities inspire those who seek to benefit from portfolio optimization and wish to advance its frontiers.

In this review, written to celebrate the 50th Anniversary of the Association of European Operational Research Societies (EURO), we have aimed to provide balanced coverage of theoretical and methodological contributions to portfolio optimization. These contributions are universal; in our view, there are no major differences in how they are being leveraged in different parts of the world. However, in our coverage of applications–which by necessity is representative rather than comprehensive–we have given some precedence to European researchers and journals when making inevitable choices due to space

\* Corresponding author. E-mail addresses: ahti.salo@aalto.fi (A. Salo), mdoumpos@tuc.gr (M. Doumpos), juuso.liesio@aalto.fi (J. Liesiö), kzopounidis@tuc.gr (C. Zopounidis).

https://doi.org/10.1016/j.ejor.2023.12.031

Received 15 August 2023; Accepted 27 December 2023 Available online 29 December 2023

0377-2217/© 2024 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



limitations. We note that many European researchers are engaged both in methodological development and real applications, which fosters the fruitful interplay between theory and practice, thereby mitigating the risk of formulating unrealistic models that fail to meet the demands encountered in practice.

In terms of scope, we have purposely abstained from covering computational techniques of machine learning and artificial intelligence (AI) for portfolio optimization. On the one hand, this is because these techniques are covered in recent reviews such as Gunjan and Bhattacharyya (2023). On the other hand, while these techniques have become increasingly pervasive in finance, they are less relevant in selecting portfolios of indivisible real assets for which relevant data from the past tends to be more limited. Furthermore, most of our emphasis is mainly on what problem features can be incorporated into portfolio optimization models. For the technical details of solving these models, we refer to recent references which also include solution-oriented reviews (see, e.g., Loke et al., 2023).

The rest of this paper is structured as follows. Section 2 gives an overview of the historical origins of portfolio optimization. Section 3 outlines the main elements of optimization models for portfolio selection. Section 4 considers optimization approaches, and Section 5 presents examples from the expansive literature on applications. In Section 6, we share our perspectives on opportunities for future research and practice.

### 2. The origins of portfolio optimization

For this paper, *portfolio optimization* is defined at the juncture of its two constituent terms, *portfolio* and *optimization*. Historically, the etymological roots of the term *portfolio* can be traced to the Italian term *portafoglio* which refers to a case or wallet containing documents such as drawings by an artist. Over the years, this term has taken on broader meanings. In business and finance, a *portfolio* refers to any collection of assets that can be considered investments. We adopt this definition, noting, however, that the assets in the portfolio need not be financial, nor does their value have to be measured in monetary terms. For instance, built infrastructures, R&D projects, and patents constitute assets. Importantly, though, because these assets are *investments*, they entail decision opportunities that can be pursued by committing resources with the aim of receiving later benefits.

*Optimization* uses mathematical methods to help identify the most preferred decision alternative out of all alternatives that satisfy relevant constraints, based on a systematic evaluation with regard to the criteria that reflect the decision-maker's preferences. Consequently, portfolio optimization is concerned with alternative *portfolios of assets* to support the selection of the *preferred portfolio* consisting of those assets that contribute most to the attainment of stated objectives, based on an evaluation with regard to evaluation criteria for these objectives while satisfying the constraints that apply. There can be different kinds of constraints, both on what properties the individual assets in the portfolio may have to fulfil and how the selected portfolio is required to perform. Typically, there are uncertainties because, for instance, the benefits gained by committing resources to the selected assets are uncertain at the time of portfolio selection.

In view of the above, we define portfolio optimization as the use of mathematical methods to support the selection of preferred portfolios of assets by accounting for the decision-maker's preferences, relevant constraints and uncertainties. This definition is generic as it does not refer to specific application fields such as finance or R&D management. An important rationale for the present paper–which discusses the breadth of portfolio optimization in many contexts–is to highlight similarities across application areas so that synergies between these can be exploited.

Historically, though, the literature on portfolio optimization has evolved in parallel tracks, propelled by the specific characteristics in application domains where, on the one hand, investment decisions can be meaningfully represented by *continuous* decision variables (e.g., choosing the share of stocks for a financial investment portfolio) or by *discrete* variables (e.g., making 'lumpy' investments such as the construction of bridges). Portfolio optimization problems with discrete variables can be addressed with portfolio decision analysis (PDA; e.g., Liesiö et al., 2021; Salo et al., 2011). One of the key application areas of PDA is project portfolio management, which covers all the activities in selecting, monitoring, and managing project portfolios (see, e.g., Levine, 2005). Thus, there is plenty of overlap between methods for project portfolio selection (for a review, see Kandakoglu et al., 2023) and PDA. Yet the scope of PDA is broader in that the alternatives need not be 'projects' in a conventional sense. Moreover, the aims of PDA may go beyond selection decisions to support the design of multi-stage decision processes, for instance.

Next, we give a brief overview of the evolution of portfolio optimization, assuming that the reader has some knowledge of decision analysis, risk management, finance, and mathematical optimization.

# 2.1. Financial portfolios

The founding principles of the modern portfolio optimization theory were laid in the seminal work of Markowitz (1952) who considered the construction of portfolios of financial assets. This theory effectively assumes that investors seek to maximize the expected utility of the portfolio's return. Markowitz observed that various utility functions can be approximated by a quadratic function of an investment's expected return and variance (Markowitz, 2014). This leads to the well-known mean–variance model which can be formulated as the bi-objective quadratic optimization problem

$$\mathbf{v} - \max_{\lambda_1, \dots, \lambda_m \in \mathcal{A}} \quad (f_1, f_2) = \left(\sum_{j=1}^m \lambda_j \mu_j, -\sum_{j=1}^m \sum_{j'=1}^m \lambda_j \lambda_{j'} \sigma_{j,j'}\right)$$
Subject to:  

$$\lambda_1 + \lambda_2 + \dots + \lambda_m = 1,$$
(1)

where  $\lambda_j$  is the proportion of capital allocated to asset j,  $\mu_j$  is the expected return of asset j,  $\sigma_{j,j'}$  is the covariance of the returns of assets j and j', and A is the set of admissible capital allocations.

Starting from the 1960s, the foundations of the mean-variance model received much attention concerning the generality of this normative theory and its consistency with the classical decision-making axioms of von Neumann and Morgenstern. Specifically, early studies in this area characterized the conditions under which the twoparameter mean-variance model can be used to analyse the choices of an expected-utility-maximizing investor (see, for instance, Hakansson, 1972; Samuelson, 1967; Tobin, 1969). Later studies presented empirical results to evaluate the quality of the approximations derived by using the mean-variance quadratic formulation to different types of utility functions (Levy & Markowitz, 1979; Pulley, 1981). While such studies have, generally, provided positive results, Samuelson (1970) showed that the mean-variance model is most useful when risk is low. In contrast, the consideration of higher moments provides improved results in the general case. Moreover, Kane (1982) noted that under a discretetime portfolio rebalancing scheme, the mean-variance model may be a poor approximation for expected utility, and examined the properties of an enhanced model that considers skewness as an additional parameter.

Beyond studies on the foundations of the mean–variance model, the basis for alternative measures of risk beyond variance was laid in the 1970s and 1980s. Specifically, the theory of stochastic dominance was introduced in the context of portfolio selection (Bawa, 1978). The foundations of downside risk measures were established (Fishburn, 1977), which spawned an active research area on coherent risk measures over the next decades (Artzner et al., 1999). Moreover, new risk measures were introduced, particularly measures that can be optimized through linear programming formulations such as the Gini mean difference model of Yitzhaki (1982) and the mean-absolute deviation model of Konno and Yamazaki (1991). Further early developments that have impacted the evolution of portfolio optimization include the formulation of multi-period and dynamic optimization approaches (Hakansson, 1971; Karatzas et al., 1987) as well as the introduction of the multicriteria and multiobjective methodologies for portfolio selection (Colson & De Bruyn, 1989; Martel et al., 1988).

A more extensive overview of the foundations and early development of portfolio optimization theory for financial investments can be found, for instance, in Elton and Gruber (1997) and Markowitz (1999).

#### 2.2. Portfolio optimization of indivisible real assets

Although the early roots of using mathematical optimization to select a portfolio of real indivisible assets developed separately from the seminal work of Markowitz (1952), the close methodological links between these two problem types were soon recognized. That is, while (Asher, 1962) applied linear programming models to optimize the allocation of resources to a portfolio of R&D projects without referencing Markowitz's work, the survey paper of Weingartner (1966) already connected the research streams on R&D project selection, financial portfolio optimization (Markowitz, 1952), and capital rationing (Lorie & Savage, 1955). Weingartner presented the classical knapsack formulation for optimizing a portfolio of investment alternatives (projects), which can be written in the general form as

$$\max_{\lambda \in \{0,1\}^m} \sum_{j=1}^m \lambda_j v^j$$

$$A\lambda < B.$$
(2)

where  $v^j$  is the net present value (NPV) of the *j*th investment alternative, and matrix *A* and vector *B* represent relevant portfolio constraints. These constraints can include, for instance, resource constraints (e.g., budget) as well as logical dependencies between projects (mutual exclusivity of projects, contingent projects). Building on the work of Markowitz (1952), Weingartner (1966) also suggested measuring risk as the variance in the portfolio payoff, which led to the quadratic integer programming problem

$$\max_{\lambda \in \{0,1\}^m} \sum_{j=1}^m \lambda_j v^j - \alpha \sum_{j=1}^m \sum_{j'=1}^m \lambda_j \lambda_{j'} \sigma_{j,j'}$$

$$A\lambda \le B.$$
(3)

where  $\alpha$  is a predefined parameter capturing the level of risk-aversion.

In the 1980s, several articles in *European Journal of Operational Research* reported real applications of portfolio optimization models for selecting R&D projects (e.g., Golabi, 1985; Lootsma et al., 1986; Weber et al., 1990). In these applications, estimates about the project benefits were obtained by carrying out a multiattribute/criteria evaluation instead of stating these benefits as a single estimate of monetary value. Specifically, the project values in model (2) were derived from an additive multiattribute utility/value function

$$v^{j} = \sum_{i=1}^{n} w_{i} v_{i}(x_{ji}), \tag{4}$$

where  $w_i$  and  $v_i$  are the importance weight and attribute-specific value function, respectively, for the *i*th attribute and  $x_{ji}$  is the performance of the *j*th project candidate with regard to the *i*th attribute. Computationally, this extension is straightforward as the optimization problem (2) can be solved with integer linear programming (ILP). However, it is not obvious that adding up the projects' multiattribute values gives the correct overall portfolio value in the objective function. This concern was addressed by Golabi et al. (1981) who were the first to apply multiattribute value and utility theory (MAVT/MAUT; Dyer & Sarin, 1979; Keeney & Raiffa, 1976; Krantz et al., 1971) to build an axiomatic foundation for portfolio value/utility functions.

A more thorough exposition of the early developments of portfolio optimization, with an emphasis on the use of decision analytic approaches to inform choices from a discrete set of alternatives, can be found in the book on portfolio decision analysis by Salo et al. (2011).

# 3. Dimensions of portfolio optimization problems

To highlight the diversity of portfolio optimization problems, we next elaborate on the salient dimensions of portfolio problems. As such, these dimensions are useful in that they can be helpful, for example, in assessing how similar different problems are.

#### 3.1. Divisibility of assets

The decision opportunities afforded by assets may vary depending on whether the investments are *divisible* (i.e., it is possible to make very small investments that are meaningfully represented by continuous decision variables) or *indivisible* (i.e., the investments are lumpy, as is the case with go/no-go projects).

In a typical setting for optimizing portfolios of financial assets, the composition of the portfolio is defined by asset weights  $\lambda_j$  which indicate what proportion of the available capital is invested in the *j*th asset. If the assets are perfectly divisible, all allocations with  $0 \le \lambda_j \le 1$  are admissible (assuming that there are no possibilities for taking a short position, in which case  $\lambda_j$  would be negative). Investments into real assets typically correspond to decisions in which the asset is either included in or excluded from the portfolio. This can be represented by the binary variable  $\lambda_j \in \{0, 1\}$ . In the portfolio optimization problem, there may be both divisible and indivisible assets at the same. This gives rise to hybrid settings where, for instance, a company is considering investments into financial instruments along with capital investments into real assets.

#### 3.2. Decision-makers and stakeholders

There may be one or several DMs whose preferences need to be represented in the portfolio optimization model, as exemplified by the case studies Vilkkumaa, Salo et al. (2014) and Fasth et al. (2020). In addition, there may be stakeholders who are not empowered to act as decision-makers but whose values and interests need to be accounted for by developing multicriteria models for decision support (for an overview, see Salo et al., 2021).

#### 3.3. Decision criteria

The performance of a portfolio, as measured by its contribution to the attainment of decision objectives, may involve one or more decision criteria. For instance, net present value (NPV) is often employed as a criterion in assessing the attractiveness of projects that are expected to be profitable. In recent years, non-financial factors have gained considerable interest, exemplified by environmental, social, and governance criteria for socially responsible investments (Ballestero et al., 2012; Utz et al., 2015). Typically, several criteria are needed to capture the DMs' preferences.

#### 3.4. Planning horizon

Portfolio optimization problems differ with regard to the length of the time horizon over which the relevant decision consequences of the portfolio are considered. Problems with longer time horizons tend to involve more uncertainty, leading to questions about how these uncertain future consequences, such as later benefits, will be considered in decision-making. A portfolio may also contain assets with different maturities. In this case, all the consequences for some assets may be realized before the end of the planning horizon, which may give rise to a reinvestment problem. It is also possible that not all investment opportunities are available or known at the beginning of the planning horizon. As a result, there is a need to decide what share of available resources is committed to the presently available investment opportunities, on the one hand, and how much is reserved for the investment opportunities that may be available later, on the other hand.

# 3.5. Timing of decisions

Portfolio problems differ in terms of when it is possible to make decisions that change the portfolio's composition, for instance, by investing in further assets, altering the level of earlier investments, or discontinuing investments (see, e.g., Angelelli et al., 2008; Gustafsson & Salo, 2005; Vilkkumaa et al., 2015). Here, one can distinguish between single-period, multi-period, and continuous portfolio optimization problems. These categories are not mutually exclusive as there may be assets about which decisions can be made on a continuous basis, whilst decisions about others can be made only at specific points in time.

In finance, dynamic optimization models (i.e., multi-period and continuous) help design portfolio management strategies that account for transaction costs and changing market conditions such as price trends. Moreover, the multi-period setting employs discrete-time models in which investment decisions can be taken only at specific points in time while continuous portfolio optimization assumes that the portfolio can be rebalanced at any time.

The intertwined decisions in the processes of asset screening (i.e., what is the set of assets to which resources can be allocated) and capital allocation (i.e., how are resources to be allocated to these assets) can be approached in different ways. These processes can be carried out in separate stages (e.g., first focusing on the evaluation of the assets and then proceeding to asset allocation; see Pendaraki et al., 2005; Xidonas et al., 2023). Alternatively, they can be considered jointly by combining both stages through integrated portfolio optimization, as illustrated by models for cardinality-constrained portfolio optimization (Bertsimas & Shioda, 2009; Woodside-Oriakhi et al., 2011). Moreover, real-world portfolio optimization is often required to provide allocations that satisfy round lot requirements (Mansini et al., 2015). In this case, the optimization result must specify the exact number of units for the investment as fractional investments are inadmissible.

#### 3.6. Uncertainties and incomplete information

There can be uncertainties about essentially all aspects of the portfolio optimization problem, including the characteristics of individual assets; their interrelationships and consequences over the planning horizon; the availability of resources that can be invested; and the preferences that guide the formulation of decision criteria and their evaluation. Here, we use the term *uncertainty* as referring to the inability to specify with certainty the value of some parameter that appears in the portfolio optimization model, such as the decision consequences associated with selecting a given portfolio.<sup>1</sup>

Uncertainties can be addressed through several approaches, most notably by using standard probability theory to model parameters as random variables or by characterizing the lack of complete information by placing bounds on what parameter values are viewed as possible. This latter approach leads to an optimization model that accommodates incomplete information and can be explored, for example, to derive robust decision recommendations, which are justified in view of all parameter values within the stated bounds. Overall, there is a broad range of approaches to modelling uncertainties, including the use of theories with non-additive expected utilities (see, e.g., Gilboa, 2009)

Importantly, uncertainties give rise to *risk* in that the realized consequences of the portfolio *ex-post* may be worse than what was expected *ex-ante* when the portfolio was selected. Overall, risk is a central concept in investment decisions, as evidenced by the wide range of risk measures and the burgeoning literature on coherent risk measures (Artzner et al., 1999; Lim et al., 2011; Szegö, 2002). Beyond

the traditional use of variance as a measure of risk, popular risk measures include semi-variance (Markowitz et al., 1993), higher-order risk measures (e.g., skewness and kurtosis, Kerstens et al., 2011; Kim et al., 2014), mean absolute deviation (Konno & Yamazaki, 1991), drawdown measures (Chekhlov et al., 2005), value-at-risk (Jorion, 2006) and conditional value-at-risk (Rockafellar & Uryasev, 2002). Examples of further financial performance measures include stochastic dominance criteria (Ogryczak & Ruszczyński, 1999; Roman et al., 2006), the Omega ratio (Kapsos, Zymler et al., 2014), the Gini mean difference index (Mansini et al., 2014), as well as risk-adjusted performance indicators (e.g., Sharpe, Sortino, etc.).

Even if the uncertain estimates about the consequences of assets in the optimization model are unbiased, the performance of the selected portfolio *ex-post* will, on average, fall short of what one would expect based on the *ex-ante* estimates. Thus, just like problems of choosing one of many alternatives, portfolio optimization problems exhibit the optimizer's curse phenomenon (Smith & Winkler, 2006) in that those assets whose performance has been overestimated tend to have a higher chance of being selected. This phenomenon can be mitigated by employing systematic debiasing techniques (see, e.g., Kettunen & Salo, 2017; Vilkkumaa, Liesiö et al., 2014) or robust satisficing (Long et al., 2022).

The consequences of a portfolio may be influenced by factors such as the future direction of the economy that are uncertain when the portfolio is selected. Here, one may distinguish between *exogenous* uncertainties associated with random variables that do not depend on the selected portfolio; and *endogenous* uncertainties which are impacted by this portfolio. From the modelling perspective, exogenous uncertainties are easier to accommodate as there is no causal relationship between them and the portfolio selection. For instance, Contingent Portfolio Programming (Gustafsson & Salo, 2005) helps select project portfolios whose consequences depend on scenarios whose probabilities are not impacted by project selections. In the case of endogenous uncertainties, in contrast, there is a need to characterize these causal relationships by specifying conditional probabilities that depend on the selections (see, e.g. Salo et al., 2022; Vilkkumaa et al., 2018). This may call for a significant elicitation effort.

Established by researchers working on decision-aiding methodologies, the dominance-based rough set approach supports portfolio decisions by permitting comparisons between probability distributions and considering the investor's preferences over time (Greco et al., 2010, 2013). There are also further approaches for representing uncertainties and varying degrees of belief, most notably fuzzy set theory. In this paper, however, we do not cover approaches based on fuzzy numbers, partly because these cannot be readily integrated with probabilistic representations that are compatible with statistical methods and the expected utility theory. Recent advances in using non-additive expected utility theories for portfolio selection are discussed by Baker et al. (2020). An overview of alternative representations of uncertainty in multicriteria problems is given in Durbach and Stewart (2012).

### 3.7. Constraints

Depending on the problem, portfolio optimization models may feature many kinds of constraints:

- Logical constraints represent logical relationships that the selected assets in a feasible portfolio must fulfil. Such constraints arise, for instance, in selecting projects of which some may be prerequisites for others.
- Resource constraints arise from limitations concerning the availability of different types of resources, such as the size of the available budget, that must be committed to implement the selected portfolio.
- *Risk constraints* place bounds on the degree of variability that the consequences of feasible portfolios are allowed to have.

<sup>&</sup>lt;sup>1</sup> We do not consider uncertainties that pertain to non-mathematical aspects such as whether or not all relevant stakeholders have been identified and consulted in formulating the portfolio model.

• *Strategic constraints* represent requirements concerning the contribution the selected portfolio has to make on one or more strategically important decision criteria. They may also represent requirements on the composition of a portfolio to ensure, for example, an acceptable balance across different kinds of assets. In financial portfolios, strategic constraints may also limit the number of different kinds of assets in the portfolio (cardinality-constrained portfolios), round lots, transaction costs, or other elements that are integral to the investment process.

One can distinguish between *hard* constraints, which must not be violated, and *soft* constraints, which may be violated to some extent, albeit at a penalty. Moreover, the availability of resources need not be given exogenously, as the completion of some projects may generate resources that can be deployed to make subsequent investments into other assets (see, e.g., Champion et al., 2023).

#### 4. Solution approaches for portfolio optimization problems

### 4.1. Bi-objective optimization

In finance, bi-objective models for exploring the risk-return tradeoff have attracted the most attention. The general form of these models can be expressed in two alternative ways:

$$\begin{array}{c|c} \min & \operatorname{Risk}(\lambda) \\ \text{s.t.:} & \operatorname{Return}(\lambda) \ge \rho \\ \lambda \in \mathcal{A} \end{array} \end{array}$$

$$\begin{array}{c|c} (5) \\ \lambda \in \mathcal{A} \end{array}$$

$$\begin{array}{c} \max & \operatorname{Return}(\lambda) - \alpha \operatorname{Risk}(\lambda) \\ \text{s.t.:} & \lambda \in \mathcal{A} \end{array} \right\}$$

$$(6)$$

where  $\operatorname{Return}(\lambda)$  and  $\operatorname{Risk}(\lambda)$  are the return and risk functions defined by the asset weights  $\lambda = (\lambda_1, \dots, \lambda_m)$ . In formulation (5), the risk-return trade-off is defined through the return constraint  $\operatorname{Return}(\lambda) \ge \rho$ , where  $\rho$  denotes the expected return that the investor seeks to achieve. On the other hand, in formulation (6), the risk weight  $\alpha > 0$  represents the relative importance of risk in the objective.

Different efficient portfolios can be obtained by changing the parameters  $\rho$  and  $\alpha$ . If the set A is convex, both formulations trace out all efficient portfolios as the parameters  $\rho$  in (5) and  $\alpha$  in (6) are varied (but there is no direct correspondence between the portfolios that are generated for different values of these two parameters). However, if the set A is non-convex, the formulation (5) outperforms the weighted model (6) in the sense that the latter may fail to identify unsupported efficient portfolios.

The exact formulation of the general forms (5) and (6) depends on how the return and risk functions are defined as well as the other dimensions of the portfolio optimization problem as outlined in Section 3. The resulting models are expressed as linear, quadratic, or non-linear optimization problems whose structure depends very much on the adopted risk measure. For instance, while variance is a quadratic function of assets' weights, other risk measures, such as the mean absolute deviation and conditional value-at-risk, can be written in a linear form. More complex risk measures, such as skewness and kurtosis, lead to non-linear optimization problems. Some constraints may lead to mixed-integer optimization problems. For instance, imposing constraints on the number of kinds of assets requires the introduction of binary variables  $\mathbf{z} = (z_1, \dots, z_m)$  to indicate whether an asset is included in the portfolio or not, and the addition of the constraints  $\lambda \geq \epsilon \mathbf{z}$  and  $z_1 + \dots + z_m \leq K$ , where  $\varepsilon$  is the minimum weight for the assets in the portfolio and K is the pre-specified maximum number of assets in the portfolio.

Portfolio optimization problems with non-linear or combinatorial features give rise to computational challenges. This has motivated the development of a wide range of special solution algorithms, heuristics, and evolutionary approaches. Algorithms based on analytical approaches are effective for solving problems with a few hundred

assets (Cesarone et al., 2013; Graham & Craven, 2021). Bertsimas and Cory-Wright (2022) present an algorithm based on the cuttingplane method that scales up well for larger instances, too. Heuristics and meta-heuristics, too, have been extensively used for finding highquality portfolio allocations. Maringer (2005) presents the main contributions and application contexts for such algorithms, whereas Erwin and Engelbrecht (2023) provide a comprehensive list of various algorithms and an overview of the literature covering the past three decades.

Another strong line of research has focused on the uncertainty of the inputs required to implement portfolio optimization models. Optimization under uncertainty can be addressed through two main paradigms: stochastic programming and robust optimization. In stochastic programming, uncertainty is described through probability distributions, permitting the generation of solutions that satisfy constraints in a probabilistic sense. Robust optimization, on the other hand, is more appropriate when there is a lack of knowledge about probabilities and a need to consider possible realizations of the uncertain parameters in a pre-specified set (Bertsimas et al., 2011). Moreover, robust optimization models are computationally easier to solve.

Instead of relying on point estimates for input parameters (e.g., expected returns and covariances), the robust optimization framework assumes that the inputs belong to uncertainty sets around the point estimates. For instance, a simple box uncertainty set  $\mathcal{U} = \{\mu \mid |\mu_i - \hat{\mu}_i| \leq$  $\delta_i$  assumes that the vector of unknown returns  $\mu = (\mu_1, \dots, \mu_m)$ belongs to an area around the vector of estimates  $\hat{\mu}$ , which is typically defined through historical data. The parameters  $\delta = (\delta_1, \dots, \delta_m)$  define the size of the uncertainty set. However, this straightforward way of defining the uncertainty set focuses solely on returns and ignores the correlations between the assets. Other types of uncertainty sets have been used to address this shortcoming, such as ellipsoidal and mixture distribution uncertainty sets (Fabozzi et al., 2010). The framework of robust optimization is applicable in the context of various bi-objective models based on different risk-return measures, beyond the mean-variance model, such as the Sharpe ratio (Chakrabarti, 2021), value-at-risk (Ghaoui et al., 2003; Zhu & Fukushima, 2009), and the Omega ratio (Kapsos, Christofides et al., 2014). Multi-period models have also been proposed (Gülpınar & Rustem, 2007; Ling et al., 2020). A bibliographic overview of this area for financial portfolios can be found in Xidonas et al. (2020).

Recently, there has been growing interest in distributionally robust optimization (DRO) which combines elements from both stochastic and robust optimization. DRO captures uncertainties about outcomes (Wiesemann et al., 2014) by assuming that the probability distribution belongs to an ambiguity set. This leads to results that are less conservative than those of other robust approaches (Goh & Sim, 2010). Mohajerin Esfahani and Kuhn (2018) and Postek et al. (2016) develop computationally tractable formulations for DRO problems under mild assumptions. They also apply the DRO approach to portfolio optimization using the conditional value-at-risk risk measure. Chen et al. (2023) present a similar approach by employing using mean absolute deviation as a measure of risk.

The frameworks of robust optimization and DRO are mainly datadriven in that they consider uncertainties about model parameters on which there is relevant historical data or which can be simulated. Nevertheless, the formulation of portfolio optimization models often requires that information about the decision-maker's preferences is also specified. This information may also involve many kinds of uncertainties (e.g., incompleteness, vagueness, etc.). We elaborate on this area of research in Sections 4.3 and 4.4.

#### 4.2. Multi-objective approaches

The multi-faceted nature of portfolio selection calls for a sufficiently comprehensive framework that spans multiple decision criteria for characterizing the performance of alternative portfolios. Specifically, approaches that account for multiple conflicting objectives help investors and portfolio managers explore different investment options in view of their preferences. Their relevance for portfolio optimization has been discussed extensively by Aouni et al. (2018) and Steuer et al. (2007).

Multiobjective optimization (MOO) models incorporate different investment decision criteria in identifying efficient (i.e., non-dominated) portfolios. The MOO theory provides solution procedures based on  $\epsilon$ constraint formulations, scalarising functions, and goal programming (GP) (for an overview of such approaches, see Miettinen, 1999). Each of these approaches transforms a MOO problem into a single-objective formulation that can be solved through standard solvers. In the  $\varepsilon$ constraint approach, one of the objectives is optimized, while imposing constraints on the others. A simple example is the bi-objective riskreturn model (5) which has been applied to financial portfolio optimization by Xidonas and Mavrotas (2014). The  $\varepsilon$ -constraint approach works well with a few objectives (i.e., two or three), but identifying the efficient frontier becomes more complex in many-objective problems. This limitation is addressed by other approaches, including those that use scalarising functions to combine the objectives into a single measure to be optimized.

The simplest scalarising function is the weighted sum. As noted in our elaboration of Eq. (1), a major limitation of weighted sum formulations is that they may not identify all efficient portfolios when the optimization model is non-convex due to integer-valued decision variables, for example. Other options are also available, such as approaches that focus on identifying portfolios that minimize deviations from reference values (points) specified by the decision-maker to define the investment objectives. An example of such a scalarising function using the maximum values  $f_1^*, \ldots, f_n^*$  of the objectives as the reference point (assuming the objectives are in a maximization form) is the augmented weighted Chebyshev metric

$$\max_{i=1...,n} [w_i(f_i(\lambda) - f_i^*)] + \theta \sum_{i=1}^n (f_i(\lambda) - f^*)$$

where  $w_1, \ldots, w_n \ge 0$  are the weights associated with the objectives and  $\theta$  is a small positive constant for ensuring that the minimization will not suggest a weakly efficient portfolio. Pavlou et al. (2019) employ this approach to construct financial portfolios combining three performance criteria: return, mean absolute deviation, and conditional value-atrisk. Dächert et al. (2022) use a similar Chebychev metric for strategic asset allocation for an insurance company.

GP formulations are also based on optimizing the deviations from a set of predefined target values on the decision criteria. Compared with scalarisation models, GP provides additional flexibility in modelling and treating deviations (e.g., the prioritization of the goals), formulating the investor's goals, and specifying the corresponding target levels. For instance, Ballestero et al. (2012) build a GP model to combine financial and environmental, social and governance (ESG) goals, whereas Tamiz et al. (2013) and Tamiz and Azmi (2019) propose an extended framework that accounts for goals related to the macroeconomic environment, regional factors, as well as accounting and stock market performance ratios. A review of related GP applications can be found in Aouni et al. (2014) and Colapinto et al. (2019).

MOO approaches, such as those mentioned above, can, in principle, be applied in an *a priori* or *a posteriori* setting (Hirschberger et al., 2013; Hwang & Masud, 1979). In the first case, the investor's preferences are incorporated in a MOO model to obtain a unique solution, i.e., the portfolio that best matches the investor's preferences subject to the constraints on the investment policy. A central part of these approaches is the modelling of preferences, for example, by developing decision models based on value/utility function representations, outranking relations, or decision rules. Sub-Section 4.3–4.6 discuss these approaches.

A posteriori implementations, on the other hand, focus on the identification of all non-dominated portfolios from which the investor can then choose the most preferred one. This calls for efficient computational procedures for MOO. As with approaches for the bi-objective case, analytical and algorithmic methodologies have been proposed. In the MOO setting, analytical methods for portfolio optimization do not incorporate complex non-linearities or combinatorial features. For instance, Oi and Steuer (2020), Oi et al. (2017) derive the analytical algebraic solution for the efficient surface in portfolio optimization problems with at least three objectives, assuming that variance is the main measure of risk. For complex instances that require the solution of computationally intensive models, multiobjective evolutionary algorithms (MOEAs) have been extensively used. Reviews by Ponsich et al. (2013) and Metaxiotis and Liagkouras (2012) on the applications of MOEAs in financial portfolio optimization indicate that such approaches are typically used to address the complexity due to non-linear performance measures and the introduction of realistic constraints. Commonly, there are two to five objectives that are considered as performance measures. Except for return and variance, these objectives include value-at-risk, expected shortfall, social responsibility, skewness, and the Sharpe ratio. The constraints mainly focus on the composition and diversification of the portfolios (e.g., cardinality constraints, bounds on the asset weights), and transactions-related issues (e.g., transaction costs, round lots, turnover constraints).

As in the case of bi-objective optimization approaches, handling uncertainty has attracted interest in the MOO context. Based on the framework for bi-objective models, the first framework for data-driven robust optimization can be found in Fliege and Werner (2014) who introduce the robust Pareto frontier and illustrate how it can be applied to portfolio optimization using an ellipsoidal uncertainty set. MOO formulations based on worst-case scenarios have also been considered. For instance, Xidonas et al. (2017) use a minimax regret criterion to model uncertainties concerning the investor's risk-return attitude (i.e., the weights of risk-return criteria). Caçador et al. (2020) use a similar minimax regret model to consider the uncertainty with respect to the covariance matrix of asset returns, combined with a genetic algorithm to solve the resulting non-linear optimization model.

An alternative path to address uncertainties in multi-objective models for portfolio optimization has arisen in stochastic programming. For example, Abdelaziz et al. (2007) present a chance-constrained compromise programming formulation to model the uncertainties the returns of financial assets and their systematic risk ( $\beta$  coefficient). Masmoudi and Ben Abdelaziz (2017) present a similar approach while Aouni et al. (2013) employ a stochastic GP approach to model the portfolio selection process for venture capital investments, using criteria related to the return and risk of the investment, as well as the survival rate and intellectual capital rate. Bravo et al. (2010) employ a stochastic GP for the expected utility of portfolio returns.

#### 4.3. Incomplete information on multiattribute preferences

Multiattribute portfolio models require the specification of several parameters that reflect the decision-maker's preferences. For instance, using the additive multiattribute value function (4) to produce project values  $v^1, \ldots, v^m$  requires that the attribute weights  $w_1, \ldots, w_n$  and the attribute-specific performances (scores)  $v_i(x_{ji})$  are elicited for each investment candidate *j*. Arguably, such an elicitation process may require a significant cognitive effort, which makes it difficult to ensure that the parameter estimates are reliable.

Here, preference disaggregation and learning techniques can be useful. Disaggregation methods use techniques such as ordinal regression to infer decision models that are as consistent as possible with holistic judgements provided by the decision-maker (Jacquet-Lagrèze & Siskos, 2001). These approaches are not restricted to the elicitation of value function models as they can be applied together with various decision models, including outranking relations and decision rules (cf. Section 4.6). Pendaraki et al. (2005) employ such an approach to infer an additive value function model for evaluating and selecting mutual funds in order to construct fund-of-funds portfolios. Ehrgott et al. (2009) use a similar approach in an integrated framework in which a value function is first inferred from the investor's holistic judgements, followed by constructing the value-maximizing portfolio through optimization.

In a typical disaggregation setting, the inference process yields point estimates for the parameters of a decision model. However, these estimates involve uncertainties due to the subjectiveness in the inputs provided by the decision-maker (i.e., holistic judgements). It is, therefore, advisable to complement decision recommendations based on the identification of the optimal portfolio with global sensitivity analyses focusing on how the optimal portfolio would change in response to changes in the parameter values.

These concerns have motivated the development of portfolio optimization approaches that explicitly capture incomplete information about multiattribute preferences. Technically, these approaches often rely on set inclusion in which parameters are constrained to a set of feasible values that is constructed from the stated preference information. Decision recommendations are then generated based on the identification of non-dominated portfolios, i.e., those feasible portfolios for which there does not exist another feasible portfolio that has a greater value for all allowed parameter values, or decision rules (e.g., maximin, minimax-regret). For instance, Liesiö et al. (2007) consider a set of attribute weights  $W \subset \mathbb{R}^n$ , defined through a system of linear constraints corresponding to the decision-makers' preference statements and intervals  $v_j(x_{ji}) \in [\underline{v}_{ji}, \overline{v}_{ji}]$  for the investments alternatives' attribute-specific values (scores). Arguably, allowing the scores to take on any values within these intervals leads to conservative decision recommendations, which has motivated the development of models that limit the number of decision alternatives whose scores deviate from their most likely value in the spirit of robust optimization (Fliedner & Liesiö, 2016; Hassanzadeh et al., 2014).

Vilkkumaa, Salo et al. (2014) extend the set inclusion approach to group decision-making by modelling incomplete information about the relative importance of the decision-makers in the group as well as the individual preferences of each group member. Liesiö and Punkka (2014) consider incomplete information on the baseline value that specifies the value of not selecting an investment alternative into the portfolio (Clemen & Smith, 2009; Golabi et al., 1981). Set inclusion has also been used to capture incomplete information about the costs of investment alternatives through interval-valued coefficients in the portfolio constraint  $A\lambda \leq B$  in model (2) (Liesiö et al., 2008; Lourenço et al., 2012). Furthermore, portfolio optimization helps guide the allocation of resources to decision-making units (DMUs) whose performance in converting these inputs to outputs can be examined with efficiency analysis. A key insight from such analyses is that allocating more resources to DMUs with high efficiency scores while withdrawing resources from those with low scores may lead to a less efficient portfolio of DMUs, suggesting that resource allocation decisions should be approached through portfolio optimization rather than extrapolation from efficiency analysis (Liesiö et al., 2020a).

Computationally, many of the above approaches lead to a multiple objective zero-one linear programming (MOZOLP) problem from which the set of non-dominated portfolios is computed. When the investment alternatives' attribute-specific consequences or costs are interval-valued, the optimization problem has interval-valued objective function coefficients. Exact algorithms for this problem by Liesiö et al. (2007, 2008) are based on the algorithms of Villarreal and Karwan (1981) for solving MOZOLP problems with point-estimate objective function coefficients. The exact algorithm by Argyris et al. (2011) can identify the set of those non-dominated portfolios that are optimal for some attribute weights (cf. supported efficient solutions). Approximate algorithms for the interval-valued MOZOLP problem have been developed by Mild et al. (2015). Challenges in identifying all non-dominated solutions have also motivated the development of heuristic evolutionary algorithms (Doerner et al., 2006; Gutjahr et al., 2010) as well as interactive methods that identify some of the non-dominated portfolios based on interactive process eliciting decision-maker's preference information (see, e.g., Hassanzadeh et al., 2014).

#### 4.4. Incomplete information on risk-preferences and state probabilities

The decision-maker's risk attitude needs to be modelled with an appropriate risk measure or a utility function to incorporate risk aversion into portfolio optimization. However, the specification of the exact functional form for the utility function may be time-consuming because there is a need to complete a lengthy process consisting of a series of preference elicitation questions. Thus, it can be helpful to provide decision recommendations that can be justified in view of different assumptions about risk preferences (e.g., a range of risk-aversion levels).

In this setting, the concept of stochastic dominance (SD) helps compare the outcome distributions for decision alternatives (in our case portfolios of assets), making it possible to determine whether or not one of them is preferred to the other for every utility function belonging to a specific class of functions. Concretely, if the outcome distribution dominates another in the sense of first-order stochastic dominance (FSD; Quirk & Saposnik, 1962), the former portfolio has a higher expected utility than the latter for any increasing utility function. Second-order stochastic dominance (SSD; Hadar & Russell, 1969) holds if the expected utility is higher for all increasing and concave utility functions. SSD is particularly relevant for portfolio optimization because it is compatible with the preferences of risk-neutral and risk-averse decision-makers.

The most popular approach for incorporating stochastic dominance into portfolio optimization is to introduce constraints to ensure that the optimal portfolio dominates a predefined benchmark distribution (e.g., return distribution of the market portfolio in case of financial assets). Kuosmanen (2004) develop MILP models to handle FSD constraints. Because the incorporation of SSD constraints does not require binary decision variables, it leads to LP models as demonstrated by Dentcheva and Ruszczynski (2003), Post (2003), and Kuosmanen (2004). More recently, Post and Kopa (2013) and Kallio and Dehghan Hardoroudi (2019) propose constraints for higher orders of stochastic dominance to model preferences consistent with smaller subsets of increasing concave utility functions. Portfolio optimization models with stochastic dominance constraints under incompletely specified state probabilities are developed by Dupacova and Kopa (2014) and Liesiö et al. (2020b).

When incompletely defined risk preferences are accounted for through stochastic dominance constraints, there is a need to define a benchmark portfolio (or a distribution) that the optimal portfolio must dominate. Yet the resulting decision recommendations can be sensitive to the selection of this benchmark. This issue could be circumvented by identifying all feasible portfolios that are not stochastically dominated, albeit this would be computationally more challenging than the incorporation of stochastic dominance constraints. Still, identifying all stochastically non-dominated portfolios by solving multiple objective optimization problems has attracted attention. Liesiö and Salo (2012) and Vilkkumaa et al. (2018) solve all non-dominated portfolios of indivisible investment alternatives in view of incomplete probability and risk-preference information in scenario planning. More general approaches for identifying the set of non-dominated portfolios by solving multiobjective optimization problems have been developed by Liesiö et al. (2023).

#### 4.5. Non-additive multiattribute value and utility functions

In practice, the additive-linear portfolio value function is the most widely used preference model for optimizing portfolios of indivisible investment alternatives evaluated with regard to multiple attributes. In this function, the portfolio value is expressed as the sum of the multiattribute values of the alternatives that are included in the portfolio. Formally, substituting the additive multiattribute-value/utility function (4) into the objective function of portfolio optimization problem (2) yields the linear-additive portfolio optimization model

$$\max_{\substack{\lambda \in \{0,1\}^m \\ A \lambda \leq B.}} \sum_{i=1}^m \lambda_j \sum_{i=1}^n w_i v_i(x_{ji}),$$
(7)

It is rather straightforward to use this model. To elicit preferences, it suffices to specify an additive multiattribute value function that aggregates the consequences of each investment alternative into a single-dimensional value. Moreover, the optimal portfolio can be identified with any standard zero–one linear programming (or ILP) algorithm. The potential shortcoming of model (7) is that it cannot capture non-linearities in the overall portfolio value. For instance, the model assumes that an alternative's additional contribution to this portfolio value is the same (i.e., constant), regardless of the other alternatives the portfolio contains.

These concerns have motivated research on non-additive portfolio value/utility functions. Already Golabi et al. (1981) proposed that the multiplicative multiattribute utility function should be employed (instead of the additive utility function) to model the utilities of individual investments in (7); this still preserves the ZOLP structure of the optimization problem. Grushka-Cockayne et al. (2008) develop nonadditive portfolio value functions and corresponding MILP formulations to support selecting a portfolio of improvement actions in air traffic management. Argyris et al. (2014) consider preferences represented by any concave multiattribute value function and develop MILP formulations to build portfolio optimization approaches in which the decision-maker's preferences are interactively elicited. Liesiö (2014) and Liesiö and Vilkkumaa (2021) establish axiomatic foundations for families of symmetric multi-linear portfolio value and utility functions. They also develop tailored implicit enumeration algorithms as well as ILP and MILP formulations to identify the optimal portfolio under such value/utility functions.

#### 4.6. Other preference models

While value/utility function models have been used extensively to model preferences in portfolio optimization, a smaller strand of literature developed primarily by European researchers has explored other approaches to preference modelling, such as outranking methods and dominance-based decision rules. The main advantage of these approaches is that they are flexible in modelling preference structures that do not fit the standard normative framework of utility theory.

The first application of outranking models to portfolio optimization was presented by Martel et al. (1988) on using ELECTRE I and II methods to evaluate financial portfolios based on the comparison of their risk-return profiles. Concretely, the ELECTRE I method identifies portfolios that best match the decision-maker's preferences, whereas ELECTRE II ranks the portfolios from the best to the worst. The ELEC-TRE I method has been employed by Perez Gladish et al. (2007) as a component of a three-stage methodology for financial portfolio selection that combines multivariate statistical techniques for modelling returns, a fuzzy multiobjective optimization model for portfolio construction, and the ELECTRE I method for portfolio selection. Apart from portfolio selection, outranking models have supported the initial screening of assets, which is an integral part of the portfolio optimization process. For example, Xidonas et al. (2009) employ the ELECTRE Tri method to sort stocks into performance categories.

Approaches to guide investments into real indivisible assets have also been proposed based on preference models incorporating outranking relations. For instance, Vetschera and de Almeida (2012) present optimization models in which the principles of the PROMETHEE V method are adapted to construct project portfolios. Kandakoglu et al. (2022) combine PROMETHEE methods both with Stochastic Multicriteria Acceptability Analysis (SMAA, Lahdelma et al., 1998) and an integer Table 1

Recent examples of applications of financial portfolio optimizat	ion.
--	------

Field	Authors	Methodology
Mutual funds	Tamiz et al. (2013)	GP
	Utz et al. (2014)	Inverse MO
	Zhou et al. (2018)	DEA
Indices	Abid et al. (2023)	GP
Bonds	Li (2019)	DP
	Drenovak et al. (2021)	MOEA
Cryptocurrencies	Brauneis and Mestel (2019)	MV
	Hashemkhani Zolfani et al. (2022)	MCDA
	Maghsoodi (2023)	MCDA
	Youssef et al. (2023)	GP
Commodities	Al Janabi et al. (2017)	NLP
	Wang et al. (2022)	MV
Derivatives	Zymler et al. (2013)	RO
	Fu et al. (2014)	DP
	Gülpınar and Çanakoğlu (2017)	RO
Insurance	Dächert et al. (2022)	MO

CP: compromise programming, DP: dynamic programming, FMO: fuzzy multiobjective optimization, GP: goal programming, MCDA: multicriteria decision analysis, MINLP: mixed-integer non-linear programming, MO: multiobjective optimization, MOEA: multiobjective evolutionary algorithm, NLP: non-linear programming, MV: mean-variance, RO: robust optimization.

programming model to generate a cluster of maintenance projects under resource constraints. Balderas et al. (2022) and Fernandez et al. (2015) present hybrid methodologies that integrate outranking approaches founded on ELECTRE methods with multi-objective portfolio optimization models.

Greco et al. (2013) propose a further hybrid approach to financial portfolio optimization. Using the dominance-based rough set approach (DRSA), they capture investors' risk-return preferences and identify efficient portfolios with the mean–variance model. The resulting decision rules from DRSA serve as inputs to an iterative and interactive optimization process that identifies the portfolio that best matches the investors' preferences. DRSA has also been applied to optimize portfolios of real assets. For example, Barbati et al. (2018) formulate a multi-objective optimization problem whose criteria indicate how many projects in the portfolio attain specified reference levels, whereafter DRSA is used to determine the optimum portfolio.

#### 5. Applications, empirical results, and discussion

To a significant extent, portfolio optimization has evolved through efforts to tackle challenges in building and solving models for real-life applications. In effect, the literature on the applications of portfolio optimization is so voluminous that we cannot review it comprehensively. Rather, we seek to offer insights based on reported applications and recent review papers.

#### 5.1. Financial portfolios

Most applications of portfolio optimization models for financial assets are data-driven in that the required modelling parameters are estimated from historical data. In effect, while most studies have focused on equity portfolios for which historical data are readily available, other investment contexts have also been considered. Table 1 gives a representative list of applications from the past decade that highlights the variety of assets, including mutual funds, equity indices, bonds, cryptocurrencies, commodities, derivatives, as well as insurance portfolios that have been considered in portfolio optimization. Beyond the standard mean–variance model, the solution approaches include multiobjective optimization and GP, robust optimization, dynamic and stochastic programming, multi-criteria decision analysis, and non-linear programming (for details, we refer to these papers).

While most applications in Table 1 are founded on a data-driven approach, some approaches incorporate the investor's preferences into

the analysis with a constructive scheme. For instance, Ehrgott et al. (2004) extends the risk-return framework into an enriched hierarchy of five objectives (12-month and 3-year return, annual dividend, volatility, Standard and Poor's Star Ranking) which are aggregated to form an investor-specific additive value function. The recommended portfolio allocation is then determined by maximizing the value function using metaheuristics. However, because this value function can be challenging to construct, Ehrgott et al. (2009) extend this approach by adopting the UTADIS preference disaggregation method (Doumpos & Zopounidis, 2002) in which the value function can be inferred from decision examples. Implementing these approaches can be expedited with decision support systems that provide data management capabilities as well as decision modelling and reporting tools (Samaras et al., 2008; Xidonas et al., 2021, 2011).

Such constructive approaches to financial portfolio optimization are becoming increasingly important due to the growing interest in incorporating non-financial objectives into the decision process. Many of these objectives are linked to sustainability principles such as environmental, social, and governance factors (ESG) which have been addressed through goal programming approaches (Abid et al., 2023; Ballestero et al., 2012) and multiobjective formulations (Gasser et al., 2017; Methling & von Nitzsch, 2020; Utz et al., 2014, 2015), for instance. In effect, the vast market on sustainability-linked investments creates a major opportunity for using these models to set up socially responsible investment (SRI) strategies that comply with ESG principles (see, e.g., Chen et al., 2021). These models can also be used in an explanatory framework, as noted by Utz et al. (2014). For example, the recent multiobjective formulations by Steuer and Utz (2023) suggest that the investments of ESG mutual funds fall short of their claimed sustainability policies.

Questions pertaining to the timing of portfolio decisions are addressed, for example, by Pham et al. (2022) who develop a distributionally robust portfolio optimization approach in a continuous time setting in the presence of ambiguity about expected returns as well as correlations between asset returns. Topaloglou et al. (2008) develop discrete-time models for optimizing international stock and bond portfolios through a stochastic programming approach. Because multiperiod models for portfolio optimization can pose computational challenges, advanced computational procedures and solution platforms can be useful. For instance, Östermark (2017) propose a computational approach which combines a genetic algorithm with a time-series module in a massively parallel processing platform.

While there has been plenty of interest in continuous optimization approaches, their practical usefulness has been questioned. For instance, Kolm et al. (2014) note that it can be difficult to estimate model parameters (e.g., expected returns and covariances over multiple periods), to overcome computational challenges caused by complex optimization problems, and to incorporate real-world features at large. Moreover, Elton and Gruber (1997) criticize the common assumption that returns are independent over time. They also argue that although continuous-time approaches have (mostly) confirmed the results of discrete multi-period results by employing more realistic assumptions, they have not offered new significant insights. In effect, the empirical evidence provided by Carroll et al. (2017) suggests that adjusting the composition of the portfolio will hinder performance if the transaction costs are high.

Beyond multi-period optimization schemes, trading methodologies and approaches that explicitly account for the timing of investments are also relevant. For instance, the hybrid approach of Chen and Wang (2015) combines a portfolio optimization model for capital allocation with a genetic algorithm for trading decisions. Décamps et al. (2005) develop an optimal stopping approach to help decide when to invest in a project whose current value is known but whose evolution over time is unknown. De Gennaro Aquino et al. (2023) extend the meanvariance model to a broader set of investment options in which the investor can invest in available investment opportunities or, alternatively, explore new investments that will become available in the future.

Overall, the formulation of more elaborate models and methodological approaches leads to multi-faceted questions about how these models and approaches compare with each other and, importantly, what benefits can be gained by adopting new tools as opposed to following simpler investment strategies. In response to these questions, Table 2 provides a representative list of comparative studies on portfolio optimization approaches for investments in financial assets.

Focusing on bi-objective models, DeMiguel et al. (2009) compare the naïve allocation (i.e., equal weights for all assets) with fourteen models within the mean-variance framework. Their empirical results from the US equity market show that the naïve approach is quite competitive with bi-objective optimization models in terms of its riskadjusted return (e.g., the Sharpe ratio) and turnover, particularly when the number of assets is large, there is limited historical data, and the idiosyncratic risk is low. In a similar comparative analysis, Cesarone et al. (2020) examine the stability properties of the simple risk diversification strategies and optimization approaches. They conclude that the risk parity strategy (in which the portfolio is chosen so that the assets contribute equally to the total risk) provides the most stable results. However, the risk-adjusted performance of this strategy was often inferior to optimization-based models. Georgantas et al. (2021) examine the stability of portfolio optimization models by employing robust optimization approaches. Based on an analysis of bi-objective models and their robust counterparts, they conclude that the latter has provided superior risk-adjusted performance, particularly during the 2007-2009 financial crisis. Bi-objective approaches are also considered by Mansini et al. (2003) who focus on models that can be formulated as LP problems, based on different risk-metrics such as the mean absolute deviation, the Gini mean difference (GMD) index, and conditional value-at-risk (CVaR). They report that the LP models provide more stable asset allocations than the mean-variance model, whereas GMD and CVaR performed better in their out-of-sample returns.

In a similar study, Ramos et al. (2023) cover a more extensive arsenal of LP solvable models taking transaction costs also into consideration. They employed several performance metrics in comparing the results of the optimization models with the Standard & Poor's 100 index. Although the benchmark index's performance was inferior to the portfolios constructed with optimization approaches, they note that no conclusive finding can be drawn as to the relative performance of these models when multiple performance metrics are considered. This suggests that approaches that accommodate multiple criteria can be well-aligned with the many facets in assessing portfolio performance.

Multiobjective models have been considered by Anagnostopoulos and Mamanis (2011), Ceren and Köksalan (2014), and Pavlou et al. (2019), among others. Anagnostopoulos and Mamanis (2011) compare five MOEAs in cardinality-constrained mean-variance optimization, focusing on the computational performance of algorithms and the quality of solutions. The strength Pareto evolutionary algorithm (SPEA2) performed best of the tested algorithms. Ceren and Köksalan (2014) used a data set from the Istanbul stock exchange to compare bi-objective and multiobjective models that covered four criteria (return, variance, liquidity, CVaR). They gave much attention to the conflicting nature of the objectives and the added value they bring into the portfolio optimization process. Yet a limitation of this analysis was that it was based on in-sample results.

Focusing on discrepancies between in-sample and out-of-sample performance, Pavlou et al. (2019) examined three bi-objective models and a three-objective formulation using Standard & Poor's 500 US index data. The mean–variance formulation gave the most robust results of the three bi-objective models. Nevertheless, the multiobjective model with three objectives (return, mean absolute deviation, CVaR) outperformed the bi-objective formulations in terms of its out-of-sample robustness. Table 2

Author	Objectives (beyond return)	Model type	Other features		
Mansini et al. (2003)	Variance, MAD, Maximum loss, GMD, CVaR	LP			
Angelelli et al. (2008)	MAD, CVaR	MILP	Heuristics, CCP		
DeMiguel et al. (2009)	Variance	QP			
Anagnostopoulos and Mamanis (2011)	Variance	MIQP	MOEAs, CCP		
Ceren and Köksalan (2014)	Variance, CVaR, Liquidity	LP/QP & MOO			
Pavlou et al. (2019)	Variance, MAD, CVaR	LP/QP, MOO			
Cesarone et al. (2020)	Variance, Semi-MAD, CVaR, Maximum loss, Omega ratio	LP/QP			
Georgantas et al. (2021)	Variance, CVaR, Omega ratio	LP/QP	RO		
Ramos et al. (2023)	Expected/maximum loss, Expected loss deviation, Shortfall deviation risk, Expectile VaR, Deviation expectile VaR	LP	Transaction costs, CCP		

CCP: Cardinality constrained portfolios, GMD: Gini mean difference, LP: Linear programming, MILP: Mixed integer LP, QP: Quadratic programming, VaR: Value-at-risk.

#### 5.2. Portfolio decision analysis for allocating resources to indivisible assets

The recent survey paper Liesiö et al. (2021) reviews recent developments in PDA by examining 146 papers from 2006 to 2019 with an emphasis on the features of the decision model, its solution approach, and, where applicable, the primary application area. In particular, optimization was the solution approach in 82% of the papers, highlighting the close connections between PDA and portfolio optimization. Mohagheghi et al. (2019) review the literature at the intersection of project management and project portfolio optimization. They cover 148 papers published from 1993 to 2018, based on the selection of evaluation criteria, representation of uncertainties, and choice of the solution approach. Like Liesiö et al. (2021), they conclude that the highest number of papers in this area have been published in the *European Journal of Operational Research*.

Given these comprehensive reviews on optimization models for selecting portfolios of indivisible alternatives, we do not seek to cover the wealth of research in this area. Instead, we give an overview of representative case studies to illustrate the breadth of application areas and the diversity of methodological approaches. The selection of these papers in Table 3 is geared towards relatively recent examples which better reflect state-of-the-art capabilities. For most (but not for all) of these papers, we summarize the problem and the approach taken to address it.

Defence: In their review of 54 papers, Harrison et al. (2020) point to the unique challenges in applying portfolio optimization in defence because, for example, planning horizons are often long and the consequences of deploying military systems depend on the actions of counterparties. This notwithstanding, optimization is helpful as it provides a structured framework within which relevant inputs can be synthesized to inform the design and operation of defence systems. The usefulness of such a framework is demonstrated by Kangaspunta et al. (2012) who present how combat models can be integrated with optimization models to identify portfolios of cost-effective weapons systems in achieving military goals. Davendralingam and DeLaurentis (2015), in turn, consider how the development of military capabilities in System of Systems (SoS) can be guided through robust portfolio optimization, focusing on choices from alternative architectural designs. Their approach is illustrated with a naval warfare scenario in which portfolios of viable systems are generated from a candidate list of available systems.

**R&D Management:** Gouglas and Marsh (2021) describe how the Coalition of Epidemic Preparedness Innovations (CEPI) allocated some US\$ 140M in 2017 to technology platforms to accelerate the development of effective vaccines for epidemic infectious diseases caused by some previously unknown pathogen. In this case study, 16 platform projects were assessed within a valuation framework that accounted for project-level and portfolio-level considerations. Estimates about seven factors contributing to each platform's success probability were elicited from experts and then combined using a multiplicative model. Preference information was elicited through discrete choice experiments. Finally, Monte Carlo simulation and extensive sensitivity analyses were employed to assess the performance of feasible portfolios. In retrospect, this application has been enormously impactful as six selected projects served as platforms for developing COVID-19 vaccines that helped combat the global pandemic.

Another instructive example from R&D management is given by Kloeber (2011) who gives a good overview of what approaches are commonly employed by large pharmaceutical companies. Possibilities for adopting interactive approaches are illustrated by Stummer and Heidenberger (2003) who propose a three-phase approach that starts by screening the project proposals that are worth further evaluation. In the second phase, all non-dominated portfolios are computed by solving a multiobjective integer linear programming model whose constraints include interdependencies, strategic requirements, and resource constraints. In the third phase, the decision-maker can set aspiration levels for the objectives in order to explore the set of feasible portfolios iteratively until a satisfying portfolio is identified.

**Energy and climate change**: Electricity retailers are exposed to financial risks when they procure electric energy from the spot market before delivering it to meet their customers' demands. Rocha and Kuhn (2012) show how retailers can hedge against these risks with a portfolio of electricity derivatives constructed by solving a multistage stochastic mean–variance optimization model. Computational challenges are tackled by employing linear decision rules to restrict the recourse decisions so that the optimal portfolio can be obtained from a convex quadratic program.

Fleischhacker et al. (2019) describe a case study on using portfolio optimization to promote distributed energy resources and to implement energy efficiency measures in Linz, Austria. Building on available open-source models and clustering algorithms, they equip the stakeholders with tools for calculating the capabilities and restrictions of the local energy system with regard to two partly conflicting criteria, i.e., costs

European Journal of	of O	perational	Research	318	(2024)	1 -	18
---------------------	------	------------	----------	-----	--------	-----	----

	· · · ·	0			-					
Selected	examples	of optimizing	the	allocation	of	resources	to	real	assets.	
Table 3										

Field	Authors	Methodologies
Defence	Harrison et al. (2020)	Survey of methods
	Kangaspunta et al. (2012)	Simulation, cost-efficiency analysis
	Davendralingam and DeLaurentis (2015)	Robust portfolio optimization
R&D management	Gouglas and Marsh (2021)	Stochastic dominance, Monte Carlo simulation
	Kloeber (2011)	Risk analysis, pipeline optimization
	Stummer and Heidenberger (2003)	Multiobjective optimization, aspiration levels
	Noro and Dias (2023)	Bi-objective optimization
Energy and climate change	Rocha and Kuhn (2012)	Multi-stage stochastic optimization
	Fleischhacker et al. (2019)	Scenarios, clustering algorithms
	Baker et al. (2020)	Robust PDA, belief dominance
Urban and environmental management	Lahtinen et al. (2017)	Survey of methods, application of RPM
	Fasth et al. (2020)	Conflict constrained optimization
Maintenance	Mild et al. (2015)	RPM with approximative algorithms
	Sacco et al. (2019)	RPM
	Mancuso et al. (2017)	Fault trees, Bayesian belief networks
Supply chains	Sawik (2019)	System dynamics, dynamic optimization
Built infrastructures	Barbati et al. (2023)	MCDA, PROMETHEE, ELECTRE Tri
	Roberti et al. (2017)	AHP, multiobjective optimization
Human resources	Gutjahr (2015)	Survey of methods
	Gutjahr et al. (2010)	Bi-level optimization
	Stummer and Kiesling (2009)	Multicriteria screening, aspiration levels

AHP: Analytic Hierarchy Process; RPM: Robust Portfolio Modelling; MCDA: Multicriteria Decision Analysis.

and carbon emissions; they also use scenario planning (Bunn & Salo, 1993) to examine possible lock-in effects of existing infrastructure and future developments. The results alert to the importance of the steady transformation of local energy systems in reaching economically sustainable goals.

Motivated by the recognition that long planning horizons may involve 'deep' uncertainties that cannot be captured by completely specified probability distributions, Baker et al. (2020) present an approach based on robust portfolio decision analysis to guide R&D investments into energy technologies. Specifically, they identify which portfolios of R&D investments are robust in view of the beliefs expressed by three different expert groups on how R&D investments will impact the future technological capabilities that can be leveraged to mitigate climate change. This application is also instructive as it demonstrates how portfolio optimization can be integrated with integrated assessment models (IAM).

Urban and environmental management: In general, applications of portfolio optimization for real investments typically involve multiple criteria; they may also involve investment alternatives that are rather different from each other. Focusing on environmental management, Lahtinen et al. (2017) note the need to construct portfolios consisting of many kinds of actions (e.g., energy saving measures, investments in renewables, educational projects, development and adoption of new technologies, regulation policies) so that when these actions are implemented together, they are jointly effective in contributing to multiple decision objectives such as mitigating greenhouse gas emissions, fostering biodiversity, and minimizing costs. As a result, the construction of such portfolios of actions can be usefully guided by portfolio optimization. In support of this claim, Lahtinen et al. (2017) provide pointers to several environmental management applications in which portfolio optimization techniques, such as quadratic programming and multiple objective integer optimization, have been effective in providing decision support.

Urban planning affects the quality of life within local communities over different spatial scales and time horizons. It also involves many stakeholder groups which may have conflicting interests. To manage possible conflicts, Fasth et al. (2020) construct a decision analytic framework in which stakeholders assess urban planning actions and estimate the weights of the criteria employed in this assessment. For each action, a conflict index and the overall value are calculated. All Pareto efficient action portfolios are computed from an optimization model in which different levels of conflict are treated as a constraint. Sensitivity analysis is an integral part of this framework, illustrated with data from the municipality of Upplands Vasby, Sweden. **Maintenance:** Mild et al. (2015) describe how the Finnish Transport Agency (FTA) has embraced portfolio optimization in its annual planning processes when selecting dozens of bridge maintenance projects from hundreds of project candidates. Specifically, this selection decision is guided by the Core Index values of an RPM model which accounts for multiple criteria, project interdependencies, uncertainties concerning project consequences, and financial and other relevant constraints.

Mancuso et al. (2017) employ portfolio optimization to identify cost-effective safety measures for the airlock subsystem of the CANDU nuclear power plant. Towards this end, they convert the fault tree representation of the subsystem into a Bayesian belief network as a framework for assessing the consequences of different portfolios of safety measures. In particular, they show that the optimization approach leads to considerably lower residual risk for different cost levels than the selection of safety measures based on conventional risk importance measures.

Sacco et al. (2019) develop a risk-based maintenance framework to help decision-makers select optimal maintenance plans to reduce the severity and likelihood of failures in a high-pressure natural gas pipeline in Great Britain. The results indicate, for example, that if there are no spatial constraints on budget allocation across zones, the selected maintenance actions will be concentrated only on some critical zones such as Scotland and the southernmost part of England. However, if predefined maintenance budgets are apportioned to different areas, the selected portfolio may be sub-optimal with a smaller risk reduction over the maintenance horizon.

**Supply chains:** Sawik (2019) proposes a dynamic optimization model to assist a manufacturer in choosing a portfolio of suppliers based on indicators reflecting their financial stability, production stability, product quality, and cost. The selection of suppliers and the allocation of orders to them are guided by a combined simulation-optimization framework with real-time monitoring of supply risk indicators. Specifically, a system dynamics model is first employed to assess the effect of supply risk indicators on a manufacturer's profit over a planning horizon. Then, a portfolio optimization model is solved to determine the optimal order allocation to suppliers in view of the manufacturer's risk propensity. Thus, the manufacturer can re-balance its supply portfolio in response to early changes in supply risk indicators to ultimately gain higher expected profits with lower risks.

**Built infrastructures:** Barbati et al. (2023) present an application of portfolio optimization to foster cultural heritage by prioritizing twenty project interventions on historical buildings in Naples, Italy.

Specifically, they propose an approach called Priority Based Portfolio Selection which employs multiple criteria sorting and ranking methods, such as PROMETHEE and ELECTRE TRI, to compare projects based on the consideration of qualitative and quantitative criteria. Then, feasible portfolios of projects satisfying the resource and logical constraints are identified before the final recommendations are derived through portfolio optimization. Although this approach has been initially developed for prioritizing conservation projects, it is generic enough to be applied in different contexts.

Another example related to historical buildings is presented by Roberti et al. (2017) who evaluate possible retrofit actions such as external and internal envelope insulation, airtightness improvement, windows replacement, and ventilative cooling. They use the Analytic Hierarchy Process (Saaty, 1980) to derive a conservation score for each action based on the judgements that are elicited from ten experts. This conservation score is combined with the energy needs for heating and cooling as well as thermal comfort in a multiobjective optimization model whose solution gives the optimal portfolio retrofits for a mediaeval building in Italy. This portfolio allows for a four-fold reduction in energy needs at a high thermal comfort level.

Human resources: Gutjahr et al. (2010) develop a multi-objective optimization model to guide the selection of projects considering both economic benefits, on the one hand, and the impacts that alternative project assignments have on the evolution of competencies, on the other hand. Pareto-optimal solutions with regard to these two criteria are obtained by decomposing the problem into (i) a master problem for portfolio selection and (ii) a slave problem for assigning personnel to the work packages of selected projects. Linearized formulations are provided to solve the slave problem efficiently, and experimental results are presented for synthetically generated test instances as well as for real data from a software-intensive organization. Related themes in managing human resources are addressed by Stummer and Kiesling (2009) who view human capital not only as an input resource required for conducting research but also as the output of pursuing that research. Specifically, they present a multi-criteria decision support system (MCDSS) with financial and non-financial objectives and constraints. In this system, the set of Pareto-efficient solutions is first determined so that the decision-maker can interactively explore and filter this set. More generally, Gutjahr (2015) reviews project selection models that incorporate skills development by learning and/or forgetting.

#### 5.3. Discussion

The above summary of applications demonstrates the breadth of problems that can be tackled through portfolio optimization. Still, one may ask what kinds of decision problems are particularly amenable to portfolio optimization and what caveats one should be aware of in seeking to benefit from it.

For starters, portfolio optimization has been immensely impactful in finance. Its profusion has been expedited by good access to data resources and, in many cases, apparent clarity concerning relevant decision criteria (e.g., expected return, volatility). Moreover, because major financial commitments are at stake, considerable benefits can be expected by moving from a sub-optimal portfolio to the optimal one. It is also usually possible to *validate* the benefits of portfolio optimization *ex-post*, for example by examining the financial outcomes and associated consequences for the decision-maker that would have materialized if the investment decision had been different from the one recommended by portfolio optimization. Still, even if there is abundant historical data on financial assets, the financial environment continues to evolve dynamically, which limits the value of the information that back-testing validation procedures provide for future investment decisions (Reschenhofer et al., 2020).

An example of another field with influential penetration of portfolio optimization is pharmaceutical R&D (cf. Gouglas & Marsh, 2021) in

which projects advance through a well-defined stage-gate structure, guided by evaluation criteria that focus on the assessment of financial aspects and the demonstration of safety, and efficacy (see Hesarsorkh et al., 2021; Kloeber, 2011). Yet in the selection of 'far-out' innovation projects, it is often much less clear what the projects are expected to achieve. Moreover, the experts may not be able to assess projects with much confidence, and it is practically impossible to ascertain *ex-post* what would have happened if another portfolio of innovation projects had been selected. These challenges may explain why the development of advanced quantitative methods for selecting innovation projects has had a limited impact on practice in settings where the innovation process is unstructured, unpredictable, and ambiguous (Si et al., 2022).

In view of the above, it is easier to justify the use of portfolio optimization in problems in which there are concrete benefits from introducing a structured decision process with a well-defined time horizon, clarity about the investment alternatives and the ability to elaborate a sufficiently comprehensive set of decision criteria with regard to which the alternatives can be evaluated with a reasonable level of confidence. Conversely, if significant difficulties are encountered in building a validated optimization model, then using numerical optimization results for making investment choices may not be warranted. Still, attempts to build such a model may help expose, for example, limitations in data resources or conflicting perceptions about the 'right' problem scope. An awareness of such obstacles can be instructive in suggesting preparatory activities that need to be completed before a useful optimization model can be built. When making efforts towards this end, one should remain attentive to any concerns that may have been inadvertently neglected or purposely omitted in specifying the model.

Many benefits of portfolio optimization stem from its ability to provide recommendations for portfolio selection based on a systematic comparison within a broad set of comparable alternatives. Intuitively, one might think that choosing this portfolio of 'best' alternatives from a truly comprehensive set of alternatives would be better than pooling the results for optimizing smaller subsets containing these same alternatives. Yet several reasons suggest that the temptation to extend the scope of portfolio models in terms of time horizons or different kinds of alternatives should be resisted:

- If the set from which a portfolio is selected contains very different kinds of investment alternatives, it may be harder to specify tangible evaluation criteria that can be meaningfully interpreted and consistently applied across all these alternatives. Therefore, depending on the context, one may aspire to make comparisons across alternatives that are sufficiently 'similar' so that they can be legitimately compared.
- In many cases, extending the time horizon over which the portfolio is optimized will lead to more uncertainties about the consequences of the investment alternatives, which may erode the credibility of the results. Also, the ensuing reinvestment problem must be recognized and addressed if the assets are not of the same maturity.
- Constructing an optimization model in order to select the portfolio from a very large set of alternatives may involve indirect costs, for example, due to greater administrative efforts or the need to postpone decisions on some alternatives to create such a set. Thus, it is pertinent to assess if the benefits of optimizing for this comparatively larger set outweigh the costs of building such a set in the first place. Furthermore, when new investment alternatives arise continually, the decision process needs to be designed in view of expectations concerning what future alternatives may become available. For example, Vilkkumaa et al. (2015) present multi-stage simulation–optimization models to support the reallocation of resources to new alternatives.

Models of portfolio optimization can be built to tackle new problems or to establish decision-making practices in which such models are solved recurrently, even routinely, with the aim of gaining enduring benefits. Developing and adopting such practices can be seen as a learning process which may take time. Yet it can be expedited by distilling 'lessons learned' from using portfolio models for decision support (see, e.g., Mild et al., 2015) and by paying attention to their behavioural fit within the organization (Luoma, 2016).

We also stress that there are application areas in which mastering the mathematical aspects of portfolio optimization does not suffice for providing effective decision support. In environmental decisionmaking (Lahtinen et al., 2017) and urban planning (Fasth et al., 2020), for instance, there are often stakeholder groups whose interests must be accounted for, because the recommended policy actions may be met with much resistance unless the stakeholder groups are engaged in the decision process. In such situations, competencies in the design and facilitation of participatory decision processes are crucial.

# 6. Opportunities for future research and practice

Here, we offer reflections on timely opportunities in portfolio optimization for continued research and applied work. Many of these opportunities are linked to introducing new modelling features to extend the range of problems that can be approached with portfolio optimization. Although we do not delve into technicalities, we note that many of these opportunities are linked to methodological and algorithmic challenges that need to be tackled.

- Building integrated models with a broader span of criteria and uses: In finance, concerns such as economic, social, and environmental sustainability have become increasingly important, giving rise to questions about how these kinds of inherently qualitative criteria can be best incorporated into models of portfolio optimization. This can be challenging, as the criteria must be adequately defined and the corresponding data resources must be of sufficiently high quality. Conversely, models for selecting portfolios of real assets can be extended by including financial assets and performance measures. Thus, for example, budgets need not be viewed as fixed constraints but as decision variables that can be optimized. Overall, there is much potential in developing integrated approaches, as demonstrated, for instance, by Tinoco et al. (2018) who optimize business project portfolios in view of economic, social, and environmental aspects.
- Establishing interfaces to non-numerical data sources: Much of the online qualitative textual information is relevant to portfolio optimization. For example, written comments on financial news convey information about investor sentiment. For optimization purposes, however, such information has to be quantified, which creates a need for tools capable of exploring qualitative data resources and converting such resources into numerical inputs (see, e.g., Yu et al., 2022). Such tools can also provide useful information about stakeholders' values and preferences. Thus, they extend the range of problems that can be tackled with portfolio optimization.
- Synthesizing statistical analysis with subjective perceptions of uncertainties: In many problems, there are significant uncertainties about the future consequences of alternative portfolios. Although much research has been done to improve the characterization of uncertainties, further efforts are needed to build models that accommodate not only data-driven estimates of uncertainties based on statistical techniques but also subjective views on uncertainties, obtained through structured processes of expert judgement elicitation. Incorporating such views (or beliefs, see Baker et al., 2020) may be especially useful when there is not much relevant data from the past. In pursuing such an integration, probabilistic approaches are appealing in that they make it possible to synthesize inputs from diverse information sources within the sound framework of standard probability theory.

- Optimizing decision processes for portfolio selection: It is often possible to develop an improved portfolio model through efforts that reduce uncertainties by providing more accurate estimates about the consequences of the alternative assets being considered. If these efforts are insufficient, the remaining uncertainties may be so significant that there is a high probability of choosing a suboptimal portfolio ex-post. Still, if such efforts towards reducing uncertainties consume plenty of resources, the benefits of being able to choose a better portfolio may not compensate for the resources that are consumed by these efforts. Thus, these kinds of inherent tradeoffs can be studied by employing the concept of Value of Information to design the selection process optimally in view of all stages that span from the formulation of the portfolio model to the implementation of the recommended portfolio. Specifically, the optimal design of processes for portfolio selection is thus an important topic that can be approached, for example, through simulation-optimization studies to derive well-founded triage decision rules (Keisler, 2004) that can be combined with portfolio optimization. For instance, if exceptionally excellent investment opportunities are encountered, it may be optimal to seize them promptly. In contrast, decisions on less attractive opportunities can be postponed and resolved later through portfolio optimization.
- Designing robust and new assets based on portfolio optimization: Portfolio problems often involve significant uncertainties whilst many decision-makers are risk averse. Therefore, instead of focusing on the solution that is optimal for a single choice of parameter values that proves erroneous ex-post, it may be advantageous to identify robust portfolios that perform reasonably well across a wide range of parameter values. Here, the computation of non-dominated portfolios based on the concept of stochastic dominance, for instance, is a powerful approach as one can derive conclusions about which alternative assets are included in all non-dominated portfolios and are therefore robust in view of the full range of parameter values. Although much work has been carried out in this area, these analyses can be extended to address fresh research questions in portfolio optimization. For example, rather than taking the set of alternative assets as a given, one may ask how new assets should be designed so that they will be included in many or most non-dominated portfolios.
- Ensuring the validity of models: Because portfolio optimization models rely on the information available at the time of the investment decision, the recommended portfolio is guaranteed to be optimal with respect to this information only. Thus, when the investment environment evolves, it is crucial to validate the underlying modelling hypotheses and carefully assess the selected portfolio's realized (future) performance. This kind of ex-post validation complements ex-ante approaches such as robustness analysis. It also helps build an integrated framework for establishing empirically grounded conclusions about the effectiveness of optimization models in practice. Validation can extend beyond examining a single portfolio to consider all nondominated portfolios (Kao & Steuer, 2016; Pavlou et al., 2019). Ideally, validation tests should not be limited to simple measures of profitability/return; instead, they should span a broader range of criteria in light of possible changes in the decision-makers' preferences over time.
- Enhancing human interactions: The interactions through which information is elicited from decision-makers and stakeholders, as well as the ways in which results of portfolio optimization are communicated to them are key determinants of the quality and eventual impact of decision support. For example, if the elicitation process is inadequate, the model will not be valid; and if the results are not clearly communicated, the decision recommendations may not be understood or trusted. Thus, there is a need to enhance these human interactions, for instance through new

elicitation protocols and visualization techniques. These advances need not be geared to developing very large portfolio models, which might require excessive elicitation effort. It may be more fruitful to build more parsimonious models whose results can be appreciated interactively in workshops, for instance.

While many portfolio optimization models are sophisticated for a good reason, we caution against developing models whose features are unnecessarily complex for their intended use. One reason is that the simultaneous compounding of multiple approaches in representing preferences and uncertainties, for example, may give rise to models that lack transparency. This can make it hard to explain the recommendations and, at worst, may contribute more to confusion than clarity. Yet if models fail to capture key concerns that actually matter (e.g., significant uncertainties or interdependencies between assets), there is a risk of losing validity due to an inadequate problem representation. Against this backdrop, there is a continuing need for reflective case studies with real data and real decision-makers, as only such case studies permit warranted conclusions about what kinds of portfolio models work well in practice. Such conclusions cannot be reached by speculating how imaginary decision-makers would interact with overly sophisticated or simplistic models of hypothetical problems.

#### 7. Conclusions

For a long time, portfolio optimization has been one of the most prominent fields of operational research, marked by significant theoretical, methodological, and algorithmic advances as well as compelling achievements in supporting the allocation of resources across a stunningly diverse range of problems. This celebrated history notwithstanding, much work is needed to expand its capabilities further. For example, more expressive preference models are needed to fully capture concerns such as sustainability; the impacts of unforeseen disruptions need to be accounted for in adapting risk measures for better risk management; and ever-better optimization techniques are needed to solve large portfolio models that are presently intractable. Furthermore, progress in synergistic areas such as text mining, deep learning, and human–computer interaction can be exploited to create capabilities for building, solving, and using portfolio optimization models beyond the state of play.

Overall, building on this paper's extensive scope, which covers divisible and indivisible assets, we see exciting opportunities that arise from the cross-fertilization between these complementary and synergistic areas of portfolio optimization. For example, the need to quantify and mitigate financial risks has spawned risk measures that can now be deployed to support the management of real assets. Conversely, advances in building preference models for multiple attributes can be used to address concerns such as the UN Sustainable Development Goals that transcend the scope of conventional financial evaluation criteria.

Efforts are also needed to understand even better in what kind of problems portfolio optimization has most to offer. Many of the benefits of portfolio models come from the possibility of allocating resources to alternative assets based on a more systematic and comprehensive evaluation than what would otherwise be the case. Still, if the costs of formulating such models outweigh the benefits, more straightforward approaches, such as taking decisions on individual assets sequentially one by one based on triage decision rules, may be considered. Thus, one should not take it for granted that the construction and solution of a full-blown portfolio optimization model is optimal.

Although portfolio optimization is a comparatively mature field, there are promising avenues for advancing its frontiers. These frontiers include but are not limited to incremental enhancements to its mathematical and methodological algorithmic core, because there are attractive opportunities that stem from the challenges encountered in problems that have not yet been satisfactorily mastered. In developing this emerging agenda, there is a need for a sufficiently close interplay between theory and practice to ensure pertinent aspects of validity, ranging from well-founded modelling activities to effective interaction with decision-makers. This combination of methodological rigour and practical relevance will ensure that the future of portfolio optimization will be bright indeed.

## Acknowledgement

We thank three anonymous reviewers and Jyrki Wallenius for their constructive feedback. This research was supported by the Academy of Finland (grant number 323800).

#### References

- Abdelaziz, F. B., Aouni, B., & Fayedh, R. E. (2007). Multi-objective stochastic programming for portfolio selection. *European Journal of Operational Research*, 177(3), 1811–1823.
- Abid, I., Urom, C., Peillex, J., Karmani, M., & Ndubuisi, G. (2023). PGP for portfolio optimization: application to ESG index family. *Annals of Operations Research*, in press.
- Al Janabi, M. A., Arreola Hernandez, J., Berger, T., & Nguyen, D. K. (2017). Multivariate dependence and portfolio optimization algorithms under illiquid market scenarios. *European Journal of Operational Research*, 259(3), 1121–1131.
- Anagnostopoulos, K., & Mamanis, G. (2011). The mean-variance cardinality constrained portfolio optimization problem: An experimental evaluation of five multiobjective evolutionary algorithms. *Expert Systems with Applications*, 38(11), 14208–14217.
- Angelelli, E., Mansini, R., & Speranza, M. G. (2008). A comparison of MAD and CVaR models with real features. *Journal of Banking & Finance*, 32(7), 1188–1197.
- Aouni, B., Colapinto, C., & La Torre, D. (2013). A cardinality constrained stochastic goal programming model with satisfaction functions for venture capital investment decision making. *Annals of Operations Research*, 205(1), 77–88.
- Aouni, B., Colapinto, C., & La Torre, D. (2014). Financial portfolio management through the goal programming model: Current state-of-the-art. *European Journal* of Operational Research, 234(2), 536–545.
- Aouni, B., Doumpos, M., Pérez-Gladish, B., & Steuer, R. E. (2018). On the increasing importance of multiple criteria decision aid methods for portfolio selection. *Journal* of the Operational Research Society, 69(10), 1525–1542.
- Argyris, N., Figueira, J. R., & Morton, A. (2011). Identifying preferred solutions to multi-objective binary optimisation problems, with an application to the multi-objective knapsack problem. *Journal of Global Optimization*, 49, 213–235.
- Argyris, N., Morton, A., & Figueira, J. R. (2014). CUT: A multicriteria approach for concavifiable preferences. *Operations Research*, 62(3), 633–642.
- Artzner, P., Delbaen, F., Eber, J. M., & Heath, D. (1999). Coherent measures of risk. Mathematical Finance, 9(3), 203–228.
- Asher, D. (1962). A linear programming model for the allocation of R and D efforts. IRE Transactions on Engineering Management, 9(4), 154–157.
- Baker, E., Bosetti, V., & Salo, A. (2020). Robust portfolio decision analysis: An application to the energy research and development portfolio problem. *European Journal of Operational Research*, 284(3), 1107–1120.
- Balderas, F., Fernández, E., Cruz-Reyes, L., Gómez-Santillán, C., & Rangel-Valdez, N. (2022). Solving group multi-objective optimization problems by optimizing consensus through multi-criteria ordinal classification. *European Journal of Operational Research*, 297(3), 1014–1029.
- Ballestero, E., Bravo, M., Perez-Gladish, B., Arenas-Parra, M., & Pia-Santamaria, D. (2012). Socially responsible investment: A multicriteria approach to portfolio selection combining ethical and financial objectives. *European Journal of Operational Research*, 216(2), 487–494.
- Barbati, M., Figueira, J., Greco, S., Ishizaka, A., & Panaro, S. (2023). A multiple criteria methodology for priority based portfolio selection. *Socio-Economic Planning Sciences*, 88, Article 101595.
- Barbati, M., Greco, S., Kadziński, M., & Słowiński, R. (2018). Optimization of multiple satisfaction levels in portfolio decision analysis. Omega, 78, 192–204.
- Bawa, V. S. (1978). Safety-first, stochastic dominance, and optimal portfolio choice. The Journal of Financial and Quantitative Analysis, 13(2), 255.
- Bertsimas, D., Brown, D. B., & Caramanis, C. (2011). Theory and applications of robust optimization. SIAM Review, 53(3), 464–501.
- Bertsimas, D., & Cory-Wright, R. (2022). A scalable algorithm for sparse portfolio selection. *INFORMS Journal on Computing*, 34(3), 1489–1511.
- Bertsimas, D., & Shioda, R. (2009). Algorithm for cardinality-constrained quadratic optimization. *Computational Optimization and Applications*, 43(1), 1–22.
- Brauneis, A., & Mestel, R. (2019). Cryptocurrency-portfolios in a mean-variance framework. Finance Research Letters, 28, 259–264.
- Bravo, M., Pla-Santamaria, D., & Garcia-Bernabeu, A. (2010). Portfolio selection from multiple benchmarks: A goal programming approach to an actual case. *Journal of Multi-Criteria Decision Analysis*, 17(5–6), 155–166.

- Bunn, D. W., & Salo, A. (1993). Forecasting with scenarios. European Journal of Operational Research, 68(3), 291–303.
- Caçador, S., Dias, J. M., & Godinho, P. (2020). Global minimum variance portfolios under uncertainty: A robust optimization approach. *Journal of Global Optimization*, 76(2), 267–293.
- Carroll, R., Conlon, T., Cotter, J., & Salvador, E. (2017). Asset allocation with correlation: A composite trade-off. *European Journal of Operational Research*, 262(3), 1164–1180.
- Ceren, T. Ş., & Köksalan, M. (2014). Effects of multiple criteria on portfolio optimization. International Journal of Information Technology and Decision Making, 13(01), 77–99.
- Cesarone, F., Mango, F., Mottura, C. D., Ricci, J. M., & Tardella, F. (2020). On the stability of portfolio selection models. *Journal of Empirical Finance*, 59, 210–234.
- Cesarone, F., Scozzari, A., & Tardella, F. (2013). A new method for mean-variance portfolio optimization with cardinality constraints. *Annals of Operations Research*, 205(1), 213–234.
- Chakrabarti, D. (2021). Parameter-free robust optimization for the maximum-Sharpe portfolio problem. European Journal of Operational Research, 293(1), 388–399.
- Champion, B. R., Gabriel, S. A., & Salo, A. (2023). Risk-based, multistage stochastic energy project selection. *Energy Systems*, 14, 603–640.
- Chekhlov, A., Uryasev, S., & Zabarankin, M. (2005). Drawdown measure in portfolio optimization. International Journal of Theoretical and Applied Finance, 8(1), 13–58.
- Chen, Y., & Wang, X. (2015). A hybrid stock trading system using genetic network programming and mean conditional value-at-risk. *European Journal of Operational Research*, 240(3), 861–871.
- Chen, D., Wu, Y., Li, J., Ding, X., & Chen, C. (2023). Distributionally robust meanabsolute deviation portfolio optimization using Wasserstein metric. *Journal of Global Optimization*, 87(2–4), 783–805.
- Chen, L., Zhang, L., Huang, J., Xiao, H., & Zhou, Z. (2021). Social responsibility portfolio optimization incorporating ESG criteria. *Journal of Management Science* and Engineering, 6(1), 75–85.
- Clemen, R. T., & Smith, J. E. (2009). On the choice of baselines in multiattribute portfolio analysis: A cautionary note. *Decision Analysis*, 6(4), 256–262.
- Colapinto, C., La Torre, D., & Aouni, B. (2019). Goal programming for financial portfolio management: A state-of-the-art review. *Operational Research*, 19(3), 717–736.
- Colson, G., & De Bruyn, C. (1989). An integrated multiobjective portfolio management system. Mathematical and Computer Modelling, 12(10–11), 1359–1381.
- Dächert, K., Grindel, R., Leoff, E., Mahnkopp, J., Schirra, F., & Wenzel, J. (2022). Multicriteria asset allocation in practice. OR Spectrum, 44(2), 349–373.
- Davendralingam, N., & DeLaurentis, D. A. (2015). A robust portfolio optimization approach to system of system architectures. Systems Engineering, 18(3), 269–283.
- De Gennaro Aquino, L., Sornette, D., & Strub, M. S. (2023). Portfolio selection with exploration of new investment assets. *European Journal of Operational Research*, 310(2), 773–792.
- Décamps, J.-P., Mariotti, T., & Villeneuve, S. (2005). Investment timing under incomplete information. *Mathematics of Operations Research*, 30(2), 472–500.
- DeMiguel, V., Garlappi, L., & Uppal, R. (2009). Optimal versus naive diversification: How inefficient is the 1/N portfolio strategy? *The Review of Financial Studies*, 22(5), 1915–1953.
- Dentcheva, D., & Ruszczynski, A. (2003). Optimization with stochastic dominance constraints. SIAM Journal on Optimization, 14(2), 548–566.
- Doerner, K., Gutjahr, W., Hartl, R., Strauss, C., & Stummer, C. (2006). Pareto ant colony optimization with ILP preprocessing in multiobjective project portfolio selection. *European Journal of Operational Research*, 171(3), 830–841.
- Doumpos, M., & Zopounidis, C. (2002). Multicriteria decision aid classification methods. Applied optimization: vol. 73, New York: Springer.
- Drenovak, M., Ranković, V., Urošević, B., & Jelic, R. (2021). Bond portfolio management under Solvency II regulation. *The European Journal of Finance*, 27(9), 857–879.
- Dupacova, J., & Kopa, M. (2014). Robustness of optimal portfolios under risk and stochastic dominance constraints. *European Journal of Operational Research*, 234, 434–441.
- Durbach, I. N., & Stewart, T. J. (2012). Modeling uncertainty in multi-criteria decision analysis. European Journal of Operational Research, 223(1), 1–14.
- Dyer, J., & Sarin, R. (1979). Measurable multiattribute value functions. Operations Research, 27(4), 810–822.
- Ehrgott, M., Klamroth, K., & Schwehm, C. (2004). An MCDM approach to portfolio optimization. European Journal of Operational Research, 155, 752–770.
- Ehrgott, M., Waters, C., Kasimbeyli, R., & Ustun, O. (2009). Multiobjective programming and multiattribute utility functions in portfolio optimization. *INFOR: Information Systems and Operational Research*, 47(1), 31–42.
- Elton, E. J., & Gruber, M. J. (1997). Modern portfolio theory, 1950 to date. Journal of Banking & Finance, 21(11–12), 1743–1759.
- Erwin, K., & Engelbrecht, A. (2023). Meta-heuristics for portfolio optimization. Soft Computing, 27, 19045–19073.
- Fabozzi, F. J., Huang, D., & Zhou, G. (2010). Robust portfolios: Contributions from operations research and finance. Annals of Operations Research, 176(1), 191–220.
- Fasth, T., Bohman, S., Larsson, A., Ekenberg, L., & Danielson, M. (2020). Portfolio decision analysis for evaluating stakeholder conflicts in land use planning. *Group Decision and Negotiation*, 29(2), 321–343.

- Fernandez, E., Gomez, C., Rivera, G., & Cruz-Reyes, L. (2015). Hybrid metaheuristic approach for handling many objectives and decisions on partial support in project portfolio optimisation. *Information Sciences*, 315, 102–122.
- Fishburn, P. (1977). Mean-risk analysis with risk associated with below-target returns. *American Economic Review*, 67, 116–126.
- Fleischhacker, A., Lettner, G., Schwabeneder, D., & Auer, H. (2019). Portfolio optimization of energy communities to meet reductions in costs and emissions. *Energy*, 173, 1092–1105.
- Fliedner, T., & Liesiö, J. (2016). Adjustable robustness for multi-attribute project portfolio selection. European Journal of Operational Research, 252(3), 931–946.
- Fliege, J., & Werner, R. (2014). Robust multiobjective optimization & applications in portfolio optimization. European Journal of Operational Research, 234(2), 422–433.
- Fu, J., Wei, J., & Yang, H. (2014). Portfolio optimization in a regime-switching market with derivatives. *European Journal of Operational Research*, 233(1), 184–192.
- Gasser, S. M., Rammerstorfer, M., & Weinmayer, K. (2017). Markowitz revisited: Social portfolio engineering. European Journal of Operational Research, 258(3), 1181–1190.
- Georgantas, A., Doumpos, M., & Zopounidis, C. (2021). Robust optimization approaches for portfolio selection: a comparative analysis. *Annals of Operations Research*, in press.
- Ghaoui, L. E., Oks, M., & Oustry, F. (2003). Worst-case value-at-risk and robust portfolio optimization: A conic programming approach. *Operations Research*, *51*(4), 543–556.
   Gilboa, I. (2009). *Theory of decision under uncertainty*. Cambridge University, MA.
- Goh, J., & Sim, M. (2010). Distributionally robust optimization and its tractable approximations. Operations Research, 58(4-part-1), 902–917.
- Golabi, K. (1985). Selecting a portfolio of nonhomogeneous R&D proposals. European Journal of Operational Research, 21(3), 347–357.
- Golabi, K., Kirkwood, C., & Sicherman, A. (1981). Selecting a portfolio of solar energy projects using multiattribute preference theory. *Management Science*, 27(2), 174–189.
- Gouglas, D., & Marsh, K. (2021). Prioritizing investments in rapid response vaccine technologies for emerging infections: A portfolio decision analysis. *PLoS ONE*, 16(2), Article e0246235.
- Graham, D. I., & Craven, M. J. (2021). An exact algorithm for small-cardinality constrained portfolio optimisation. *Journal of the Operational Research Society*, 72(6), 1415–1431.
- Greco, S., Matarazzo, B., & Słowiński, R. (2010). Dominance-based rough set approach to decision under uncertainty and time preference. *Annals of Operations Research*, 176(1), 41–75.
- Greco, S., Matarazzo, B., & Słowiński, R. (2013). Beyond Markowitz with multiple criteria decision aiding. Journal of Business Economics, 83(1), 29–60.
- Grushka-Cockayne, Y., Reyck, B. D., & Degraeve, Z. (2008). An integrated decisionmaking approach for improving European air traffic management. *Management Science*, 54(8), 1395–1409.
- Gülpınar, N., & Çanakoğlu, E. (2017). Robust portfolio selection problem under temperature uncertainty. European Journal of Operational Research, 256(2), 500–523.
- Gülpınar, N., & Rustem, B. (2007). Worst-case robust decisions for multi-period meanvariance portfolio optimization. *European Journal of Operational Research*, 183(3), 981–1000.
- Gunjan, A., & Bhattacharyya, S. (2023). A brief review of portfolio optimization techniques. Artificial Intelligence Review, 56(5), 3847–3886.
- Gustafsson, J., & Salo, A. (2005). Contingent portfolio programming for the management of risky projects. *Operations Research*, 53(6), 946–956.
- Gutjahr, W. J. (2015). Project portfolio selection under skill development. In C. Schwindt, & J. Zimmermann (Eds.), *Handbook on project management and scheduling Vol. 2* (pp. 729–750). Cham: Springer International Publishing.
- Gutjahr, W. J., Katzensteiner, S., Reiter, P., Stummer, C., & Denk, M. (2010). Multiobjective decision analysis for competence-oriented project portfolio selection. *European Journal of Operational Research*, 205(3), 670–679.
- Hadar, J., & Russell, W. (1969). Rules for ordering uncertain prospects. The American Economic Review, 59(1), 25–34.
- Hakansson, N. H. (1971). Multi-period mean-variance analysis: Toward a general theory of portfolio choice. *The Journal of Finance*. 26(4), 857–884.
- Hakansson, N. H. (1972). Mean-variance analysis in a finite world. The Journal of Financial and Quantitative Analysis, 7(4), 1873.
- Harrison, K. R., Elsayed, S., Garanovich, I., Weir, T., Galister, M., Boswell, S., Taylor, R., & Sarker, R. (2020). Portfolio optimization for defence applications. *IEEE Access*, 8, 60152–60178.
- Hashemkhani Zolfani, S., Mehtari Taheri, H., Gharehgozlou, M., & Farahani, A. (2022). An asymmetric PROMETHEE II for cryptocurrency portfolio allocation based on return prediction. *Applied Soft Computing*, 131, Article 109829.
- Hassanzadeh, F., Nemati, H., & Sun, M. (2014). Robust optimization for interactive multiobjective programming with imprecise information applied to R&D project portfolio selection. *European Journal of Operational Research*, 238(1), 41–53.
- Hesarsorkh, A. H., Ashayeri, J., & Naeini, A. B. (2021). Pharmaceutical R&D project portfolio selection and scheduling under uncertainty: A robust possibilistic optimization approach. *Computers & Industrial Engineering*, 155, Article 107114.
- Hirschberger, M., Steuer, R. E., Utz, S., Wimmer, M., & Qi, Y. (2013). Computing the nondominated surface in tri-criterion portfolio selection. *Operations Research*, 61(1), 169–183.

- Hwang, C.-L., & Masud, A. S. M. (1979). Multiple objective decision making methods and applications. *Lecture notes in economics and mathematical systems: vol. 164*, Berlin, Heidelberg: Springer.
- Jacquet-Lagrèze, E., & Siskos, Y. (2001). Preference disaggregation: 20 years of MCDA experience. European Journal of Operational Research, 130(2), 233–245.
- Jorion, P. (2006). Value at risk: The new benchmark for managing financial risk (p. 585). McGraw-Hill, New York.
- Kallio, M., & Dehghan Hardoroudi, N. (2019). Advancements in stochastic dominance efficiency tests. European Journal of Operational Research, 276(2), 790–794.
- Kandakoglu, M., Walther, G., & Amor, S. B. (2022). A robust multicriteria clustering methodology for portfolio decision analysis. *Computers & Industrial Engineering*, 174, Article 108803.
- Kandakoglu, M., Walther, G., & Amor, S. B. (2023). The use of multi-criteria decisionmaking methods in project portfolio selection: A literature review and future research directions. *Annals of Operations Research*.
- Kane, A. (1982). Skewness preference and portfolio choice. The Journal of Financial and Quantitative Analysis, 17(1), 15–25.
- Kangaspunta, J., Liesiö, J., & Salo, A. (2012). Cost-efficiency analysis of weapon system portfolios. European Journal of Operational Research, 223(1), 264–275.
- Kao, C., & Steuer, R. E. (2016). Value of information in portfolio selection, with a Taiwan stock market application illustration. *European Journal of Operational Research*, 253(2), 418–427.
- Kapsos, M., Christofides, N., & Rustem, B. (2014). Worst-case robust omega ratio. European Journal of Operational Research, 234(2), 499–507.
- Kapsos, M., Zymler, S., Christofides, N., & Rustem, B. (2014). Optimizing the Omega ratio using linear programming. *Journal of Computational Finance*, 17(4), 49–57.
- Karatzas, I., Lehoczky, J. P., & Shreve, S. E. (1987). Optimal portfolio and consumption decisions for a "small investor" on a finite horizon. SIAM Journal on Control and Optimization, 25(6), 1557–1586.
- Keeney, R., & Raiffa, H. (1976). Decisions with multiple objectives: Preferences and value trade-offs. John Wiley & Sons, New York.
- Keisler, J. (2004). Value of information in portfolio decision analysis. *Decision Analysis*, 1(3), 177–189.
- Kerstens, K., Mounir, A., & van de Woestyne, I. (2011). Geometric representation of the mean-variance-skewness portfolio frontier based upon the shortage function. *European Journal of Operational Research*, 210(1), 81–94.
- Kettunen, J., & Salo, A. (2017). Estimation of downside risks in project portfolio selection. Production and Operations Management, 26(10), 1839–1853.
- Kim, W. C., Fabozzi, F. J., Cheridito, P., & Fox, C. (2014). Controlling portfolio skewness and kurtosis without directly optimizing third and fourth moments. *Economics Letters*, 122(2), 154–158.
- Kloeber, J. (2011). Current and cutting edge methods of portfolio decision analysis in pharmaceutical R&D. In *Portfolio decision analysis: Improved methods for resource allocation* (pp. 283–331). Springer.
- Kolm, P., Tutuncu, R., & Fabozzi, F. (2014). 60 Years of portfolio optimization: Practical challenges and current trends. *European Journal of Operational Research*, 234(2), 356–371.
- Konno, H., & Yamazaki, H. (1991). Mean-absolute deviation portfolio optimization model and its applications to Tokyo stock market. *Management Science*, 37(5), 519–531.
- Krantz, D., Luce, R., Suppes, P., & Tversky, A. (1971). Foundations of measurement, Vol. I. New York, NY: Academic Press.
- Kuosmanen, T. (2004). Efficient diversification according to stochastic dominance criteria. Management Science, 50(10), 1390–1406.
- Lahdelma, R., Hokkanen, J., & Salminen, P. (1998). SMAA stochastic multiobjective acceptability analysis. European Journal of Operational Research, 106(1), 137–143.
- Lahtinen, T. J., Hämäläinen, R. P., & Liesiö, J. (2017). Portfolio decision analysis methods in environmental decision making. *Environmental Modelling & Software*, 94, 73–86.
- Levine, H. A. (2005). Project portfolio management: A practical guide to selecting projects, managing portfolios, and maximizing benefits. John Wiley & Sons.
- Levy, H., & Markowitz, H. M. (1979). Approximating expected utility by a function of mean and variance. American Economic Review, 69(3), 308–317.
- Li, K. (2019). Portfolio selection with inflation-linked bonds and indexation lags. Journal of Economic Dynamics & Control, 107, Article 103727.
- Liesiö, J. (2014). Measurable multiattribute value functions for portfolio decision analysis. *Decision Analysis*, 11(1), 1–20.
- Liesiö, J., Andelmin, J., & Salo, A. (2020a). Efficient allocation of resources to a portfolio of decision making units. *European Journal of Operational Research*, 286(2), 619–636.
- Liesiö, J., Kallio, M., & Argyris, N. (2023). Incomplete risk-preference information in portfolio decision analysis. *European Journal of Operational Research*, 304(3), 1084–1098.
- Liesiö, J., Mild, P., & Salo, A. (2007). Preference programming for robust portfolio modeling and project selection. *European Journal of Operational Research*, 181(3), 1488–1505.
- Liesiö, J., Mild, P., & Salo, A. (2008). Robust portfolio modeling with incomplete cost information and project interdependencies. *European Journal of Operational Research*, 190(3), 679–695.

- Liesiö, J., & Punkka, A. (2014). Baseline value specification and sensitivity analysis in multiattribute project portfolio selection. *European Journal of Operational Research*, 237(3), 946–956.
- Liesiö, J., & Salo, A. (2012). Scenario-based portfolio selection of investment projects with incomplete probability and utility information. *European Journal of Operational Research*, 217(1), 162–172.
- Liesiö, J., Salo, A., Keisler, J. M., & Morton, A. (2021). Portfolio decision analysis: Recent developments and future prospects. *European Journal of Operational Research*, 293(3), 811–825.
- Liesiö, J., & Vilkkumaa, E. (2021). Nonadditive multiattribute utility functions for portfolio decision analysis. *Operations Research*, 69(6), 1886–1908.
- Liesiö, J., Xu, P., & Kuosmanen, T. (2020b). Portfolio diversification based on stochastic dominance under incomplete probability information. *European Journal* of Operational Research, 286(2), 755–768.
- Lim, A. E., Shanthikumar, J. G., & Vahn, G.-Y. (2011). Conditional value-at-risk in portfolio optimization: Coherent but fragile. *Operations Research Letters*, 39(3), 163–171.
- Ling, A., Sun, J., & Wang, M. (2020). Robust multi-period portfolio selection based on downside risk with asymmetrically distributed uncertainty set. *European Journal of Operational Research*, 285(1), 81–95.
- Loke, Z. X., Goh, S. L., Kendall, G., Abdullah, S., & Sabar, N. R. (2023). Portfolio optimization problem: A taxonomic review of solution methodologies. *IEEE Access*, 11, 33100–33120.
- Long, D. Z., Sim, M., & Zhou, M. (2022). Robust satisficing. Operations Research, 71(1), 61–82.
- Lootsma, F., Meisner, J., & Schellemans, F. (1986). Multi-criteria decision analysis as an aid to the strategic planning of energy R&D. European Journal of Operational Research, 25(2), 216–234.
- Lorie, J. H., & Savage, L. J. (1955). Three problems in rationing capital. Journal of Business, 28(4), 229–239.
- Lourenço, J., Morton, A., & Bana e Costa, C. (2012). PROBE a multicriteria decision support system for portfolio robustness evaluation. *Decision Support Systems*, 54(1), 534–550.
- Luoma, J. (2016). Model-based organizational decision making: A behavioral lens. European Journal of Operational Research, 249(3), 816–826.
- Maghsoodi, A. I. (2023). Cryptocurrency portfolio allocation using a novel hybrid and predictive big data decision support system. Omega, 115, Article 102787.
- Mancuso, A., Compare, M., Salo, A., & Zio, E. (2017). Portfolio optimization of safety measures for reducing risks in nuclear systems. *Reliability Engineering & System Safety*, 167, 20–29.
- Mansini, R., Ogryczak, W., & Speranza, M. G. (2003). LP solvable models for portfolio optimization: A classification and computational comparison. *IMA Journal of Management Mathematics*, 14, 187–220.
- Mansini, R., Ogryczak, W., & Speranza, M. G. (2014). Twenty years of linear programming based portfolio optimization. *European Journal of Operational Research*, 234(2), 518–535.
- Mansini, R., Ogryczak, W., & Speranza, M. G. (2015). EURO advanced tutorials on operational research, Linear and mixed integer programming for portfolio optimization. Cham: Springer.
- Maringer, D. (2005). Portfolio management with heuristic optimization. Advances in computational management science: vol. 8, Berlin/Heidelberg: Springer-Verlag.
- Markowitz, H. M. (1952). Portfolio selection. The Journal of Finance, 7(1), 77-91.
- Markowitz, H. M. (1999). The early history of portfolio theory: 1600–1960. Financial Analysts Journal, 55(4), 5–16.
- Markowitz, H. (2014). Mean-variance approximations to expected utility. European Journal of Operational Research, 234(2), 346–355.
- Markowitz, H., Todd, P., Xu, G., & Yamane, Y. (1993). Computation of meansemivariance efficient sets by the critical line algorithm. Annals of Operations Research, 45(1), 307–317.
- Martel, J.-M., Khoury, N. T., & Bergeron, M. (1988). An application of a multicriteria approach to portfolio comparisons. *Journal of the Operational Research Society*, 39(7), 617–628.
- Masmoudi, M., & Ben Abdelaziz, F. (2017). A chance constrained recourse approach for the portfolio selection problem. Annals of Operations Research, 251(1–2), 243–254.
- Metaxiotis, K., & Liagkouras, K. (2012). Multiobjective evolutionary algorithms for portfolio management: A comprehensive literature review. *Expert Systems with Applications*, 39(14), 11685–11698.
- Methling, F., & von Nitzsch, R. (2020). Tailor-made thematic portfolios: A core satellite optimization. Journal of Global Optimization, 76(2), 317–331.
- Miettinen, K. (1999). Nonlinear multiobjective optimization. Dordrecht: Springer.
- Mild, P., Liesiö, J., & Salo, A. (2015). Selecting infrastructure maintenance projects with robust portfolio modeling. *Decision Support Systems*, 77, 21–30.
- Mohagheghi, V., Mousavi, S. M., Antucheviciene, J., & Mojtahedi, M. (2019). Project portfolio selection problems: A review of models, uncertainty approaches, solution techniques, and case studies. *Technological and Economic Development of Economy*, 25(6), 1380–1412.
- Mohajerin Esfahani, P., & Kuhn, D. (2018). Data-driven distributionally robust optimization using the wasserstein metric: Performance guarantees and tractable reformulations. *Mathematical Programming*, 171(1–2), 115–166.

- Noro, J., & Dias, L. C. (2023). Project portfolio management considering the commitment of agents: A bi-objective model applied to administrative services. *Journal of* the Operational Research Society, 74(4), 1049–1062.
- Ogryczak, W., & Ruszczyński, A. (1999). From stochastic dominance to mean-risk models: Semideviations as risk measures. *European Journal of Operational Research*, 116(1), 33–50.
- Östermark, R. (2017). Massively parallel processing of recursive multi-period portfolio models. *European Journal of Operational Research*, 259(1), 344–366.
- Pavlou, A., Doumpos, M., & Zopounidis, C. (2019). The robustness of portfolio efficient frontiers. *Management Decision*, 57(2), 300–313.
- Pendaraki, K., Zopounidis, C., & Doumpos, M. (2005). On the construction of mutual fund portfolios: A multicriteria methodology and an application to the greek market of equity mutual funds. *European Journal of Operational Research*, 163(2), 462–481.
- Perez Gladish, B., Jones, D., Tamiz, M., & Bilbao-Terol, A. (2007). An interactive three-stage model for mutual funds portfolio selection. *Omega*, 35(1), 75–88.
- Pham, H., Wei, X., & Zhou, C. (2022). Portfolio diversification and model uncertainty: A robust dynamic mean-variance approach. *Mathematical Finance*, 32(1), 349–404.
- Ponsich, A., Jaimes, A. L., & Coello, C. A. C. (2013). A survey on multiobjective evolutionary algorithms for the solution of the portfolio optimization problem and other finance and economics applications. *IEEE Transactions on Evolutionary Computation*, 17(3), 321–344.
- Post, T. (2003). Empirical tests for stochastic dominance efficiency. The Journal of Finance, 58(5), 1905–1931.
- Post, T., & Kopa, M. (2013). General linear formulations of stochastic dominance criteria. European Journal of Operational Research, 230(2), 321–332.
- Postek, K., den Hertog, D., & Melenberg, B. (2016). Computationally tractable counterparts of distributionally robust constraints on risk measures. SIAM Review, 58(4), 603–650.
- Pulley, L. B. (1981). A general mean-variance approximation to expected utility for short holding periods. *The Journal of Financial and Quantitative Analysis*, 16(3), 361–373.
- Qi, Y., & Steuer, R. E. (2020). On the analytical derivation of efficient sets in quadand-higher criterion portfolio selection. Annals of Operations Research, 293(2), 521-538.
- Qi, Y., Steuer, R. E., & Wimmer, M. (2017). An analytical derivation of the efficient surface in portfolio selection with three criteria. *Annals of Operations Research*, 251(1-2), 161–177.
- Quirk, J. P., & Saposnik, R. (1962). Admissibility and measurable utility functions. *Review of Economic Studies*, 29(2), 140–146.
- Ramos, H. P., Righi, M. B., Guedes, P. C., & Müller, F. M. (2023). A comparison of risk measures for portfolio optimization with cardinality constraints. *Expert Systems* with Applications, 228, Article 120412.
- Reschenhofer, E., Kaur Mangat, M., Zwatz, C., & Guzmics, S. (2020). Evaluation of current research on stock return predictability. *Journal of Forecasting*, 39(2), 334–351.
- Roberti, F., Oberegger, U. F., Lucchi, E., & Troi, A. (2017). Energy retrofit and conservation of a historic building using multi-objective optimization and an analytic hierarchy process. *Energy and Buildings*, 138, 1–10.
- Rocha, P., & Kuhn, D. (2012). Multistage stochastic portfolio optimisation in deregulated electricity markets using linear decision rules. *European Journal of Operational Research*, 216(2), 397–408.
- Rockafellar, R., & Uryasev, S. (2002). Conditional value-at-risk for general loss distributions. Journal of Banking & Finance, 26(7), 1443–1471.
- Roman, D., Darby-Dowman, K., & Mitra, G. (2006). Portfolio construction based on stochastic dominance and target return distributions. *Mathematical Programming*, 108(2–3), 541–569.
- Saaty, T. L. (1980). The analytic hierarchy process: Planning, priority setting, resource allocation. McGraw-Hill.
- Sacco, T., Compare, M., Zio, E., & Sansavini, G. (2019). Portfolio decision analysis for risk-based maintenance of gas networks. *Journal of Loss Prevention in the Process Industries*, 60, 269–281.
- Salo, A., Andelmin, J., & Oliveira, F. (2022). Decision programming for mixedinteger multi-stage optimization under uncertainty. *European Journal of Operational Research*, 299(2), 550–565.
- Salo, A., Hämäläinen, R. P., & Lahtinen, T. J. (2021). Multicriteria methods for group decision processes: An overview. In D. M. Kilgour, & C. Eden (Eds.), *Handbook* of group decision and negotiation (pp. 863–891). Cham: Springer International Publishing.
- Salo, A., Keisler, J., & Morton, A. (Eds.), (2011). Portfolio decision analysis: Improved methods for resource allocation. International series in operations research & management science: vol. 162, Springer.
- Samaras, G. D., Matsatsinis, N. F., & Zopounidis, C. (2008). A multicriteria DSS for stock evaluation using fundamental analysis. *European Journal of Operational Research*, 187, 1380–1401.
- Samuelson, P. A. (1967). General proof that diversification pays. The Journal of Financial and Quantitative Analysis, 2(1), 1–13.
- Samuelson, P. A. (1970). The fundamental approximation theorem of portfolio analysis in terms of means, variances and higher moments. *Review of Economic Studies*, 37(4), 537–542.
- Sawik, T. (2019). Disruption mitigation and recovery in supply chains using portfolio approach. Omega, 84, 232-248.

- Si, H., Kavadias, S., & Loch, C. (2022). Managing innovation portfolios: From project selection to portfolio design. *Production and Operations Management*, 31(12), 4572–4588.
- Smith, J. E., & Winkler, R. L. (2006). The optimizer's curse: Skepticism and postdecision surprise in decision analysis. *Management Science*, 52(3), 311–322.
- Steuer, R. E., Qi, Y., & Hirschberger, M. (2007). Suitable-portfolio investors, nondominated frontier sensitivity, and the effect of multiple objectives on standard portfolio selection. *Annals of Operations Research*, 152, 297–317.
- Steuer, R. E., & Utz, S. (2023). Non-contour efficient fronts for identifying most preferred portfolios in sustainability investing. *European Journal of Operational Research*, 306(2), 742–753.
- Stummer, C., & Heidenberger, K. (2003). Interactive R&D portfolio analysis with project interdependencies and time profiles of multiple objectives. *IEEE Transaction on Engineering Management*, 50(2), 175–183.
- Stummer, C., & Kiesling, E. (2009). A multicriteria decision support systems for competence-driven project portfolio selection. *International Journal of Information Technology and Decision Making*, 8(2), 379–401.
- Szegö, G. (2002). Measures of risk. *Journal of Banking & Finance*, *26*(7), 1253–1272. Tamiz, M., & Azmi, R. A. (2019). Goal programming with extended factors for portfolio.
- Faliniz, M., & Azini, R. A. (2019). Goal programming with extended factors for portional selection. International Transactions in Operational Research, 26(6), 2324–2336.
- Tamiz, M., Azmi, R. A., & Jones, D. F. (2013). On selecting portfolio of international mutual funds using goal programming with extended factors. *European Journal of Operational Research*, 226(3), 560–576.
- Tinoco, M. A. C., Dutra, C. C., Ribeiro, J. L. D., Miorando, R. F., & Caten, C. S. T. (2018). An integrated model for evaluation and optimisation of business project portfolios. *European Journal of Industrial Engineering*, 12(3), 442–463.
- Tobin, J. (1969). Comment on Borch and Feldstein. Review of Economic Studies, 36(1), 13-14.
- Topaloglou, N., Vladimirou, H., & Zenios, S. A. (2008). A dynamic stochastic programming model for international portfolio management. *European Journal of Operational Research*, 185(3), 1501–1524.
- Utz, S., Wimmer, M., Hirschberger, M., & Steuer, R. E. (2014). Tri-criterion inverse portfolio optimization with application to socially responsible mutual funds. *European Journal of Operational Research*, 234(2), 491–498.
- Utz, S., Wimmer, M., & Steuer, R. E. (2015). Tri-criterion modeling for constructing more-sustainable mutual funds. *European Journal of Operational Research*, 246(1), 331–338.
- Vetschera, R., & de Almeida, A. T. (2012). A PROMETHEE-based approach to portfolio selection problems. *Computers & Operations Research*, 39(5), 1010–1020.
- Vilkkumaa, E., Liesiö, J., & Salo, A. (2014). Optimal strategies for selecting project portfolios using uncertain value estimates. *European Journal of Operational Research*, 233(3), 772–783.
- Vilkkumaa, E., Liesiö, J., Salo, A., & Ilmola-Sheppard, L. (2018). Scenario-based portfolio model for building robust and proactive strategies. *European Journal of Operational Research*, 266(1), 205–220.

Vilkkumaa, E., Salo, A., & Liesiö, J. (2014). Multicriteria portfolio modeling for the development of shared action agendas. Group Decision and Negotiation, 23(1), 49–70.

- Vilkkumaa, E., Salo, A., Liesiö, J., & Siddiqui, A. (2015). Fostering breakthrough technologies — How do optimal funding decisions depend on evaluation accuracy? *Technological Forecasting and Social Change*, 96, 173–190.
- Villarreal, B., & Karwan, M. H. (1981). Multicriteria integer programming: A (hybrid) dynamic programming recursive approach. *Mathematical programming*, 21, 204–223.
- Wang, L., Ahmad, F., Luo, G.-I., Umar, M., & Kirikkaleli, D. (2022). Portfolio optimization of financial commodities with energy futures. *Annals of Operations Research*, 313(1), 401–439.
- Weber, R., Werners, B., & Zimmermann, H.-J. (1990). Planning models for research and development. *European Journal of Operational Research*, 48(2), 175–188.
- Weingartner, H. M. (1966). Capital budgeting of interrelated projects: survey and synthesis. Management Science, 12(7), 485–516.
- Wiesemann, W., Kuhn, D., & Sim, M. (2014). Distributionally robust convex optimization. Operations Research, 62(6), 1358–1376.
- Woodside-Oriakhi, M., Lucas, C., & Beasley, J. (2011). Heuristic algorithms for the cardinality constrained efficient frontier. *European Journal of Operational Research*, 213(3), 538–550.
- Xidonas, P., Doukas, H., & Sarmas, E. (2021). A python-based multicriteria portfolio selection DSS. RAIRO - Operations Research, 55, S3009–S3034.
- Xidonas, P., Lekkos, I., Giannakidis, C., & Staikouras, C. (2023). Multicriteria security evaluation: Does it cost to be traditional? *Annals of Operations Research*, 323(1–2), 301–330.
- Xidonas, P., & Mavrotas, G. (2014). Multiobjective portfolio optimization with nonconvex policy constraints: Evidence from the Eurostoxx 50. *The European Journal* of Finance, 20(11), 957–977.
- Xidonas, P., Mavrotas, G., Hassapis, C., & Zopounidis, C. (2017). Robust multiobjective portfolio optimization: A minimax regret approach. *European Journal of Operational Research*, 262(1), 299–305.
- Xidonas, P., Mavrotas, G., & Psarras, J. (2009). A multicriteria methodology for equity selection using financial analysis. *Computers & Operations Research*, 36(12), 3187–3203.
- Xidonas, P., Mavrotas, G., Zopounidis, C., & Psarras, J. (2011). IPSSIS: An integrated multicriteria decision support system for equity portfolio construction and selection. *European Journal of Operational Research*, 210(2), 398–409.

Xidonas, P., Steuer, R., & Hassapis, C. (2020). Robust portfolio optimization: A categorized bibliographic review. *Annals of Operations Research*, 292(1), 533–552.

Yitzhaki, S. (1982). Stochastic dominance, mean variance, and Gini's mean difference. American Economic Review, 72(1), 178–185.

- Youssef, M., Naoua, B. B., Abdelaziz, F. B., & Chibane, M. (2023). Portfolio selection: Should investors include crypto-assets? A multiobjective approach. *International Transactions in Operational Research*, 30(5), 2620–2639.
- Yu, J.-R., Chiou, W. P., Hung, C.-H., Dong, W.-K., & Chang, Y.-H. (2022). Dynamic rebalancing portfolio models with analyses of investor sentiment. *International Review of Economics & Finance*, 77, 1–13.
- Zhou, Z., Xiao, H., Jin, Q., & Liu, W. (2018). DEA frontier improvement and portfolio rebalancing: An application of China mutual funds on considering sustainability information disclosure. *European Journal of Operational Research*, 269(1), 111–131.
- Zhu, S., & Fukushima, M. (2009). Worst-case conditional value-at-risk with application to robust portfolio management. *Operations Research*, 57(5), 1155–1168.
- Zymler, S., Kuhn, D., & Rustem, B. (2013). Worst-case value at risk of nonlinear portfolios. Management Science, 59(1), 172–188.